

WORKING PAPER NO. 640

An Iterative Approach to Rationalizable Implementation

Ritesh Jain, Ville Korpela, and Michele Lombardi

March 2022



University of Naples Federico II



University of Salerno



Bocconi University, Milan

CSEF - Centre for Studies in Economics and Finance DEPARTMENT OF ECONOMICS AND STATISTICS - UNIVERSITY OF NAPLES FEDERICO II 80126 NAPLES - ITALY Tel. and fax +39 081 675372 - e-mail: <u>csef@unina.it</u> ISSN: 2240-9696



WORKING PAPER NO. 640

An Iterative Approach to Rationalizable Implementation

Ritesh Jain*, Ville Korpela[†], and Michele Lombardi[‡]

Abstract

We study rationalizable implementation of social choice functions. Iterative Monotonicity is both necessary *and* sufficient for implementation when there are two or more players.

JEL classification: C79, D82.

Keywords: Implementation, iterative monotonicity, rationalizability, complete information.

Acknowledgements: This version supersedes the version of July 19, 2021, circulated under the same title. Special thanks to Siyang Xiong whose comments and suggestions have led to substantial improvements. We also thank Arunava Sen, Jeevant Rampal, Anujit Chakraborty, Takashi Kunimoto, Yi-Chun Chen and audiences at IISER (Bhopal), the University of Glasgow, IIM-Ahmedabad and SAET (2021) for useful comments. Ritesh Jain greatly acknowledges financial support from the Ministry of Science and Technology Grant, Taiwan under the grant: 107- 2410-H-001-113-MY2. All remaining errors are ours.

* Institute of Economics, Academia Sinica. Email: ritesh@econ.sinica.edu.tw

[†] Turku School of Economics. Email: vipeko@utu.fi

[‡] University of Liverpool Management School, University of Napoli Federico II, and CSEF. E-mail: michele.lombardi@liverpool.ac.uk

Table of contents

- 1. Introduction
- 2. Data and Empirical Strategy
- 3. Contextualizing The Problem
- 4. Iterative Monotonicity
- 5. Connection with The Partition Based Approach

Appendices

References

I. INTRODUCTION

Most early studies on implementation theory focused on (Bayes–)Nash equilibrium and its refinements (see, for instance, Maskin and Sjöström, 2002). These solution concepts rely on two implicit common-knowledge assumptions. (1) No strategic uncertainty: each player correctly predicts the strategic play of her opponents. (2) No higher-order uncertainty: the underlying type space, based on the set of states Θ , is assumed to be common knowledge among players. By adopting the notion of robustness of Weinstein and Yildiz (2007) in a mechanism design setting, Oury and Tercieux (2012) show that this notion is tightly connected to (full) implementation in rationalizable strategies. A social choice function (SCF) f is rationalizably implementable if there exists a mechanism such that every rationalizable strategy profile leads to the realization of f.

In a fundamental paper, Bergemann et al. (2011) (BMT) study the implementation of SCFs under complete information in rationalizable strategies. They show that a strict version of the monotonicity condition introduced by Maskin (1999) is sufficient for implementation under a no worst alternative (NWA) condition.¹ Recently, Xiong (2022) presents necessary and sufficient conditions for implementation in rationalizable strategies.

However, these studies have two significant limitations. The first is related to the fact that their conditions are not sufficient to implement two-player SCFs.² Indeed, in Section III, we construct an SCF that is Nash implementable and that satisfies the sufficient conditions of BMT, but that fails to be implementable in rationalizable strategies when there are two players. Therefore, two-player rationalizable implementation problems require a fundamentally different solution from that provided by the

¹This condition, called strict Maskin monotonicity^{*}, is also necessary under a mild restriction on the class of implementing mechanisms. NWA requires that a player never receives his worst outcome under the SCF.

²The two-person problem is an important one in the theory of incentives. Indeed, the two-player model is the leading case for contracting or bargaining applications (see, for instance, Moore and Repullo, 1990 and Dutta and Sen, 1991).

existing characterization results.

The second is related to the fact that their implementing conditions are stated in terms of the existence of a partition of the set Θ . However, no rules for how to prove or disprove its existence are given. This absence makes it challenging to conceptualize its existence from a game-theoretical standpoint. From a practical standpoint, the condition becomes difficult to check as the number of partitions of Θ grows exponentially with the size of Θ .³

Motivated by these limitations, we develop an approach to fully characterize the class of rationalizably implementable SCFs with two or more players. Our approach is constructive and is based on the idea of deceptions (Jackson (1991), Oury and Tercieux (2012)).⁴ More importantly, it allows us to solve two-player implementation problems and construct the partition specified by the existing characterization results from the primitives of the implementation model. More precisely, our necessary and sufficient condition is based on an algorithm that identifies the required partition by using the limit point of an *increasing* sequence of deceptions. The sequence has a game-theoretical interpretation. Without the guidance provided by our algorithm, one is forced to search over a larger number of partitions.⁵

Our necessary and sufficient condition for rationalizable implementation is termed Iterative Monotonicity (IM). Player *i*'s deception is a correspondence $\beta_i : \Theta \longrightarrow 2^{\Theta} \setminus \{\emptyset\}$ such that $\theta \in \beta_i(\theta)$. We denote a profile of deceptions by β and the set of all profiles by \mathcal{B}^t . The main novelty of our approach is that, for a given SCF f and a player i, we define a function \mathcal{R}_i from \mathcal{B}^t to \mathcal{B}_i^t . The self-map $\mathcal{R} \equiv \mathcal{R}_1 \times \ldots \times \mathcal{R}_I$ allows

$$B(n) = \sum_{k=0}^{n} \binom{n}{k} B(k)$$

where B(1) = 1.

³In combinatorial mathematics, the number of partitions of a set of size n is referred to as bell number. Bell numbers can be recursively defined as follows: for every n + 1,

⁴Traditionally, the idea of deceptions is used to analyze implementation problems with incomplete information. However, there are notable excepts such as Mezzetti and Renou (2017).

⁵For concreteness, when $|\Theta| = 6$, the total number of partition is 216 but our algorithm checks at most 6 partitions.

us to recursively define an increasing sequence $\{\beta_k\}_{k\geq 0}$, where $\beta_k = \mathcal{R}(\beta_{k-1})$ and the deception β^0 —where $\theta \in \beta_i^0(\theta)$ for all $\theta \in \Theta$ and every player *i*. This sequence has a clear game-theoretical interpretation, which is discussed in Section IV. The limit of this sequence which we name as β^* is the least (fixed-point) deception.⁶

An SCF f satisfies IM provided that for all $\theta, \theta' \in \Theta$, $\beta^*(\theta) \cap \beta^*(\theta')$ is empty whenever $f(\theta) \neq f(\theta')$. When f satisfies IM, β^* pins down the partition of Θ required by the existing characterization results. It is worth mentioning that IM is a measurability type condition, which is reminiscent of the classical Abreu–Matsushima measurability (Abreu and Matsushima, 1992).⁷

The rest of the paper is organized as follows. Section II presents the implementation model. Section III presents our motivating example for the two-player case. Section IV discusses our implementing condition and present our characterization result. Section V briefly concludes by connecting our analysis with the existing characterization results. Appendices include proofs not in the main body.

II. Setup

The environment consists of $I \ge 2$ players (we write $\mathcal{I} = \{1, ..., I\}$ for the set of players), a finite set of states Θ and a countable set of pure outcomes X. Let $Y \equiv \Delta(X)$ denote the set of lotteries over X. Player *i*'s preferences over lotteries are described by a utility function $u_i : Y \times \Theta \mapsto \mathbb{R}$, with

$$u_{i}(y,\theta) = \sum_{x \in X} y_{x} u_{i}(x,\theta) ,$$

where y_x is the probability of pure outcome x. For all $\theta \in \Theta$, $u_i(\cdot, \theta)$ satisfies the expected utility hypothesis. To save writing, for all $i \in \mathcal{I}$, we write -i for player i's

⁶Since \mathcal{R} is monotone (i.e., increasing) on \mathcal{B}^t and \mathcal{B}^t is a complete lattice, Tarski's fixed-point theorem implies that there is a least fixed-point of \mathcal{R} .

⁷Abreu and Matsushima (1992) proposed a measurability condition, now referred to as Abreu– Matsushima measurability, to characterize virtual rationalizable implementation when there is incomplete information.

opponents. For all $i \in \mathcal{I}$, let $\Theta_{-i} \equiv \underbrace{\Theta \times \dots \times \Theta}_{I-1\text{-times}}$, with $\theta_{-i} = (\theta_1, \dots, \theta_{i-1}, \theta_{i+1}, \dots, \theta_I)$ as a typical element of Θ_{-i} .

Given a state $\theta \in \Theta$, a player $i \in \mathcal{I}$, and a lottery $x \in Y$, the lower contour set of $u_i(\cdot, \theta)$ at x is $L_i(x, \theta) = \{y \in Y | u_i(x, \theta) \ge u_i(y, \theta)\}$; the strict lower contour set of $u_i(\cdot, \theta)$ at x is $SL_i(x, \theta) = \{y \in Y | u_i(x, \theta) > u_i(y, \theta)\}$; and the strict upper contour set of $u_i(\cdot, \theta)$ at x is $SU_i(x, \theta) = \{y \in Y | u_i(y, \theta) > u_i(x, \theta)\}$.

A mechanism \mathcal{M} is a pair $\mathcal{M} \equiv (M, g)$, where $M \equiv \prod_{i \in \mathcal{I}} M_i$, with each M_i being a nonempty countable set, and $g: M \longrightarrow Y$. As usual, we refer to M_i as the (pure) strategy space of $i \in \mathcal{I}$, to a member of M, denoted by m, as a (pure) strategy profile, and to g as an outcome function. As usual, $M_{-i} \equiv \prod_{j \in \mathcal{I} \setminus \{i\}} M_j$, with m_{-i} as a typical element. The same notational convention will be followed for any profile of objects. For all $M' \subseteq M$, let $g[M'] = \{g(m) \in Y | m \in M'\}$.

The environment, when combined with the mechanism, describes a game (of complete information) for all state $\theta \in \Theta$, which is denoted by (\mathcal{M}, θ) . We will use (correlated) rationalizability as a solution concept. Bernheim (1984) and Pearce (1984) provide a definition of rationalizability in which players' conjectures over their opponents' play are independent. In this paper, we follow the convention of some of the recent literature (e.g., Osborne and Rubinstein (1994) in using "rationalizability" for the correlated version of rationalizability (we refer the reader to Brandenburger and Dekel (1987)). Our definition of rationalizability coincides with the standard definition when strategy spaces are compact. However, our definition allows for infinite, non-compact strategy spaces. In this case, our definition is equivalent to one introduced by Lipman (1994).

Formally, let S be the set of all strategy-set profiles, defined by $S \equiv \prod_{i \in \mathcal{I}} S_i$, where $S_i \equiv 2^{M_i}$ for all $i \in \mathcal{I}$, with $S = (S_i)_{i \in \mathcal{I}}$ as a typical profile of S. The family S is a lattice with the natural ordering of the set inclusion: $S \leq S'$ if $S_i \subseteq S'_i$ for all $i \in \mathcal{I}$. The smallest element of S is denoted by $\underline{S} \equiv (\emptyset, ..., \emptyset)$, whereas the largest element is denoted by $\overline{S} \equiv M$.

Fix any game (\mathcal{M}, θ) . The strategy $m_i \in M_i$ is player *i*'s best-response to his belief $\lambda_i \in \Delta(M_{-i})$ at θ if

$$m_{i} \in \arg \max_{m'_{i} \in M_{i}} \sum_{m_{-i} \in M_{-i}} \lambda_{i} \left(m_{-i} \right) u_{i} \left(g \left(m'_{i}, m_{-i} \right), \theta \right)$$

Let us define an operator $b^{\mathcal{M},\theta} : \mathcal{S} \longrightarrow \mathcal{S}$ to iteratively eliminate never best responses with $b^{\mathcal{M},\theta} \equiv \left(b_i^{\mathcal{M},\theta}\right)_{i\in\mathcal{I}}$, where $b_i^{\mathcal{M},\theta} : \mathcal{S} \longrightarrow \mathcal{S}_i$ is defined, for all $S \in \mathcal{S}$, by

$$b_{i}^{\mathcal{M},\theta}\left(S\right) = \left\{ \begin{array}{c} m_{i} \in M_{i} \\ m_{i} \in M_{i} \end{array} \middle| \begin{array}{c} \text{there exists } \lambda_{i}^{m_{i},\theta} \in \Delta\left(M_{-i}\right) \text{ such that} \\ (1) \ \lambda_{i}^{m_{i},\theta}\left(m_{-i}\right) > 0 \implies m_{-i} \in S_{-i}, \\ (2) \ m_{i} \text{ is a best response to } \lambda_{i}^{m_{i},\theta} \text{ at } \theta \end{array} \right\}$$

Note that $b^{\mathcal{M},\theta}$ is increasing (that is, $S \leq S' \implies b^{\mathcal{M},\theta}(S) \leq b^{\mathcal{M},\theta}(S')$.

By Tarski's fixed point theorem, there exists a largest fixed point of $b^{\mathcal{M},\theta}$, which is denoted by $S^{\mathcal{M},\theta}$. That is, (1) $b^{\mathcal{M},\theta} \left(S^{\mathcal{M},\theta} \right) = S^{\mathcal{M},\theta}$ and (2) $b^{\mathcal{M},\theta} \left(S \right) = S \implies S \leq S^{\mathcal{M},\theta}$. Alternatively, the fixed point $S^{\mathcal{M},\theta}$ can be constructed by starting with the largest element of the lattice, \bar{S} , and by iteratively applying the operator $b^{\mathcal{M},\theta}$. If the strategy sets are finite, we have that

$$S_{i}^{\mathcal{M},\theta} \equiv \bigcap_{k \ge 0} b_{i}^{\mathcal{M},\theta} \left(\left[b_{i}^{\mathcal{M},\theta} \right]^{k} \left(\bar{S} \right) \right).$$

In this case, the solution concept is equivalent to deletion of strictly dominated strategies (Brandenburger and Dekel (1987)). When the strategy sets are infinite sets, transfinite induction may be necessary to reach the fixed point (Lipman (1994)). Sometimes, we will use the following notation

$$S_{i,k}^{\mathcal{M},\theta} \equiv b_i^{\mathcal{M},\theta} \left(\left[b_i^{\mathcal{M},\theta} \right]^{k-1} \left(\bar{S} \right) \right)$$

to denote the kth step of the iterative process. The set $S_i^{\mathcal{M},\theta}$ is the set of strategies

surviving (transfinite) iterated deletion of never best responses.

We refer to $m_i \in S_i^{\mathcal{M},\theta}$ as a player *i*'s rationalizable strategy in \mathcal{M} at state θ , and to a member of $S^{\mathcal{M},\theta}$ as a rationalizable strategy profile in \mathcal{M} at state θ .

We say that a profile $S \in S$ has the best-response property in state θ if $S \leq b^{\mathcal{M},\theta}(S)$, or equivalently, if for all $i \in \mathcal{I}$ and all $m_i \in S_i$, there exists $\lambda_i \in \Delta(M_{-i})$ such that $\lambda_i(m_{-i}) > 0 \implies m_j \in S_j$ for all $j \in \mathcal{I} \setminus \{i\}$, and m_i is a best-response to λ_i at θ . It can be checked that $S \leq S^{\mathcal{M},\theta}$ when S has the best-response property in state θ .

Player *i*'s mixed-strategy σ_i is a probability distribution over M_i . The space of player *i*'s mixed-strategy is denoted by Σ_i , where $\sigma_i(m_i)$ is the probability that σ_i assigns to m_i . The space of mixed-strategy profiles is denoted by $\Sigma = \prod_{i \in \mathcal{I}} \Sigma_i$, with element σ as a typical strategy profile. A mixed-strategy may assign probability one to a single strategy m_i , that is, $\sigma_i(m_i) = 1$. We refer to such a mixed-strategy as a (pure) strategy and denote it by m_i . The support of a mixed-strategy σ_i is the set of pure strategies that are played with positive probability, that is, $supp(\sigma_i) =$ $\{m_i \in M_i | \sigma_i(m_i) > 0\}$. A mixed-strategy profile σ is a Nash equilibrium of (\mathcal{M}, θ) if for all $i \in \mathcal{I}$,

$$u_{i}\left(g\left(\sigma_{i},\sigma_{-i}\right),\theta\right) \geq u_{i}\left(g\left(\sigma_{i}',\sigma_{-i}\right),\theta\right),$$

for all $\sigma'_i \in \Sigma_i$. Write $NE(\mathcal{M}, \theta)$ for the set of Nash equilibrium profiles of (\mathcal{M}, θ) , and write $g(NE(\mathcal{M}, \theta))$ for the set of Nash equilibrium outcomes of (\mathcal{M}, θ) .

An SCF f is a function $f: \Theta \longrightarrow Y$.

Definition 1. A mechanism \mathcal{M} implements $f : \Theta \longrightarrow Y$ in rationalizable strategies if for all $\theta \in \Theta$, $S^{\mathcal{M},\theta} \neq \emptyset$ and $m \in S^{\mathcal{M},\theta} \implies g(m) = f(\theta)$. If such a mechanism exists, f is said to be rationalizably implementable.

A partition of Θ is a correspondence $P : \Theta \rightrightarrows \Theta$ satisfying the following requirements: (i) $\theta \in P(\theta)$ for all $\theta \in \Theta$, (ii) $\cup_{\theta \in \Theta} P(\theta) = \Theta$, and (iii) $P(\theta) \cap P(\theta') = \emptyset$ if $P(\theta) \neq P(\theta')$. Given an SCF f, P_f is the partition of Θ induced by f, that is, $P_f = \{\Theta_y\}_{y \in f(\Theta)}$ where $\Theta_y = \{\theta \in \Theta | f(\theta) = y\}$. A partition P of Θ is at least as fine as P_f , or equivalently, P_f is coarser than P if $P(\theta) \subseteq P_f(\theta)$ for all $\theta \in \Theta$. Let \mathcal{P}_f denote the set of partitions that are at least as fine as P_f , that is,

 $\mathcal{P}_{f} = \{P | P \text{ is a partition of } \Theta \text{ such that } P(\theta) \subseteq P_{f}(\theta) \text{ for all } \theta \in \Theta \}.$

III. CONTEXTUALIZING THE PROBLEM

When there are at least three players, BMT show that Maskin monotonicity^{*} and a strengthening of the NWA condition, referred to as NWA^{*}, is sufficient for rationalizable implementation. The requirements can be stated as follows.⁸

Definition 2. $f: \Theta \to Y$ is Maskin monotonic^{*} provided that there exists a partition $P \in \mathcal{P}_f$ such that for all $\theta, \theta' \in \Theta$,

$$\left[\begin{array}{c} \text{for all } i \in \mathcal{I} \text{ and all } \hat{\theta} \in P(\theta'), \\ L_i\left(f(\theta'), \hat{\theta}\right) \subseteq L_i(f(\theta'), \theta) \end{array}\right] \implies \theta' \in P(\theta).$$

Definition 3. Suppose that $f : \Theta \to Y$ is Maskin monotonic^{*} with respect to the partition $P \in \mathcal{P}_f$. $f : \Theta \to Y$ is NWA^{*} provided that for all $(i, \theta) \in \mathcal{I} \times \Theta$, there exists $\tilde{y}_i(P(\theta)) \in Y$ such that

$$u_i(f(\theta), \theta) > u_i(\tilde{y}_i(P(\theta)), \theta).$$

BMT show that Nash implementation of f is equivalent to its rationalizable implementation when there are at least three players and f satisfies both NWA^{*} and MM^{*}. In what follows, we show below that when there are two players, this equivalence breaks down, though the constructed f satisfies both NWA^{*} and MM^{*}, moreover, it is Nash implementable. Indeed, the constructed f satisfies MM^{*} with respect to the finest partition.

Suppose that $\mathcal{I} = \{1, 2\}, X = \{a, b, c, d, e, f, g\}$ and $\Theta = \{\theta, \theta', \theta''\}$. Players' ⁸Under NWA^{*}, Maskin monotonic^{*} is equivalent to strict Maskin monotonicity^{*}. utilities from pure outcomes are summarized in the table below, where $\varepsilon \in (\frac{1}{2}, 1)$.

	$u_{1}\left(\cdot, heta ight)$	$u_{2}\left(\cdot,\theta\right)$	$u_1\left(\cdot, \theta'\right)$	$u_{2}\left(\cdot,\theta'\right)$	$u_1\left(\cdot, \theta''\right)$	$u_2\left(\cdot,\theta''\right)$
a	1	$-(1-\varepsilon)$	1	-1	1	-1
b	0	0	0	0	0	0
c	-1	1	$-(1+\varepsilon)$	1	-1	1
d	1	-2	-2	-1	1	-1
e	2	$-(2-\epsilon)$	2	-2	-2	-2
f	3	-3	-3	-3	3	-3
g	0	0	0	0	-3	-3

The planner wants to implement f, which is defined by

$$f(\theta) = f(\theta') = \{b\}$$
 and $f(\theta'') = \{a\}$.

It can easily be checked that f satisfies NWA^{*}. Moreover, it can be checked that f satisfies MM^{*} with respect the finest partition vacuously.⁹ Finally, it can be checked that f is Nash implementable.¹⁰ However, f is not rationalizable implementable. The easiest way to make this point without being distracted by boring details is to suppose that \mathcal{M} implements f in rationalizable strategies and in (pure strategy) Nash equilibria.

For each state $\bar{\theta} \in \Theta$, let $m(\bar{\theta}) = (m_1(\bar{\theta}), m_2(\bar{\theta}))$ be a Nash equilibrium strategy profile for the game $(\mathcal{M}, \bar{\theta})$. Since $SL_1(f(\theta), \theta) \cap L_2(f(\theta'), \theta')$ is nonempty, by construction, and since, moreover, \mathcal{M} rationalizable implements f, it follows that

⁹Indeed, it can be checked that $\frac{1}{2}a + \frac{1}{2}c \in L_2(f(\theta'), \theta') \cap SU_2(f(\theta'), \theta), \frac{1}{2}a + \frac{1}{2}c \in L_1(f(\theta), \theta) \cap SU_1(f(\theta), \theta'), \frac{1}{2}a + \frac{1}{2}d \in L_1(f(\theta'), \theta') \cap SU_1(f(\theta'), \theta''), e \in L_1(f(\theta''), \theta'') \cap SU_1(f(\theta''), \theta'), e \in L_1(f(\theta''), \theta'') \cap SU_1(f(\theta''), \theta'), e \in L_1(f(\theta''), \theta'') \cap SU_1(f(\theta''), \theta) \text{ and } \frac{2}{3}c + \frac{1}{3}d \in L_2(f(\theta), \theta) \cap SU_2(f(\theta), \theta'').$

¹⁰Indeed, f satisfies the necessary and sufficient condition of Moore and Repullo (1990) under the specifications that the set $B = \Delta(X)$, $C_i(f(\bar{\theta}), \bar{\theta}) = L_i(f(\bar{\theta}), \bar{\theta})$ for all $i \in \mathcal{I}$ and all $\bar{\theta} \in \Theta$, and the punishment outcomes are $e(f(\bar{\theta}), \bar{\theta}, f(\bar{\theta}), \bar{\theta}) = f(\bar{\theta})$ for all $\bar{\theta} \in \Theta$, $e(f(\theta), \theta, f(\theta'), \theta') = e(f(\theta''), \theta'', f(\theta), \theta) = g$, $e(f(\theta), \theta, f(\theta''), \theta'') = 0.1c + 0.9g$ and $e(f(\theta'), \theta', f(\theta), \theta) = e(f(\theta''), \theta', f(\theta''), \theta'') = 0.1d + 0.9g$.

 $g(m_1(\theta'), m_2(\theta)) = f(\theta)$. Since $L_1(f(\theta), \theta) \subseteq L_1(f(\theta), \theta'')$ and $L_2(f(\theta), \theta') \subseteq L_2(f(\theta), \theta'')$, it follows that $(m_1(\theta'), m_2(\theta))$ is a Nash equilibrium strategy profile for (\mathcal{M}, θ'') . However, since \mathcal{M} Nash implements f, it follows that $f(\theta) = f(\theta'')$, which is a contradiction.

f is Nash implementable because we can assign to the strategy profile $(m_1(\theta'), m_2(\theta))$ the outcome g and this allows us to satisfy the necessary and sufficient conditions for Nash implementation (Moore and Repullo, 1991). Rationalizable implementation does not allow us to make this assignment. The reason is that any mechanism that rationalizable implements f has to assign the outcome $f(\theta)$ to the rationalizable strategy profile $(m_1(\theta'), m_2(\theta))$. Therefore, when there are two players, rationalizable implementation imposes restrictions in the mechanism design exercise that are not available in the case of Nash implementation.

IV. ITERATIVE MONOTONICITY

Fix any $i \in \mathcal{I}$ and any SCF f. Let

$$\Theta_{i}^{f} = \left\{ \theta \in \Theta | SL_{i} \left(f \left(\theta \right), \theta \right) = \emptyset \right\}.$$

$$\tag{1}$$

In words, $\theta \in \Theta_i^f$ if $f(\theta)$ is the worst outcome for i at θ . Note that if $\theta \in \Theta_i^f$ and f is rationalizably implementable, every strategy of player i is a rationalizable strategy at θ .

When I = 2, for every $\theta \in \Theta$, we define $A_i(\theta)$ by

$$A_{i}(\theta) = \left\{ \theta' \middle| SL_{i}(f(\theta), \theta) \bigcap L_{-i}(f(\theta'), \theta') = \emptyset \right\} \bigcup \left\{ \theta \right\}.$$
⁽²⁾

Let us call any map $\beta_i : \Theta \longrightarrow 2^{\Theta} \setminus \{\emptyset\}$ as player *i*'s deception. A special deception for player *i* is the truth-telling deception, β_i^t , defined by $\beta_i^t(\theta) = \{\theta\}$ for all $\theta \in \Theta$.

Another special deception is denoted by β_i^0 and it is defined, for all $\theta \in \Theta$, by

$$\beta_i^0(\theta) = \begin{cases} A_i(\theta) & \text{if } I = 2\\ \beta_i^t(\theta) & \text{if } i \in \mathcal{I}^\theta \text{ and } I \ge 3\\ \Theta & \text{if } i \notin \mathcal{I}^\theta \text{ and } I \ge 3. \end{cases}$$
(3)

Note that $\theta \in \beta_i^0(\theta)$ for all $\theta \in \Theta$.

For any β_i and β'_i , we write $\beta_i \subseteq \beta'_i$ if $\beta_i(\theta) \subseteq \beta'_i(\theta)$ for all $\theta \in \Theta$. Let \mathcal{B}_i^t denote the set of player *i*'s deceptions containing the truth-telling deception, that is,

$$\mathcal{B}_{i}^{t} \equiv \left\{ \beta_{i} \in \mathcal{B}_{i} \middle| \beta_{i}^{0} \subseteq \beta_{i} \right\}.$$

$$\tag{4}$$

Let $\mathcal{B}^t \equiv \prod_{i \in \mathcal{I}} \mathcal{B}^t_i$, with $\beta = (\beta_i)_{i \in \mathcal{I}}$ as a typical deception profile of \mathcal{B}^t . For all $\beta, \beta' \in \mathcal{B}^t$, we write $\beta \subseteq \beta'$ if $\beta_i \subseteq \beta'_i$ for all $i \in \mathcal{I}$. The collection \mathcal{B}^t is a complete lattice with the natural ordering set inclusion: $\beta \leq \beta'$ if $\beta \subseteq \beta'$. The largest element is $\overline{\beta} = (\Theta, ..., \Theta)$. The smallest element is β^t .

Let us define the function $\mathcal{R}_i : \mathcal{B}^t \mapsto \mathcal{B}_i^t$ as follows, where for all $\beta \in \mathcal{B}^t$, we use $\mathcal{R}_i(\beta)$ and \mathcal{R}_i^β interchangeably. For all $\beta \in \mathcal{B}^t$, all $\theta \in \Theta$ and all $i \in \mathcal{I}$, let \mathcal{R}_i^β be defined by

$$\mathcal{R}_{i}^{\beta}(\theta) = \bigcup_{E_{i} \in 2^{\Theta} \setminus \{\emptyset\}} \left\{ \begin{array}{c} E_{i} \quad \left\{ \begin{array}{c} \text{there exists } E_{-i} \in \left(2^{\Theta} \setminus \{\emptyset\}\right)^{I-1} \text{ such that} \\ \text{for all } \left(\ell, \theta'\right) \in \mathcal{I} \times E_{\ell}, \text{ there exists} \\ \left(\tilde{\theta}, \hat{\theta}\right) \in E_{\ell} \times \bigcap_{j \in \mathcal{I} \setminus \{\ell\}} E_{j} \text{ such that} \\ \left(\theta', \hat{\theta}\right) \in \beta_{\ell} \left(\tilde{\theta}\right) \times \bigcap_{j \in \mathcal{I} \setminus \{\ell\}} \beta_{j} \left(\tilde{\theta}\right) \text{ and} \\ either \ \tilde{\theta} \in \Theta_{\ell}^{f} \text{ or } L_{\ell} \left(f\left(\hat{\theta}\right), \hat{\theta}\right) \subseteq L_{\ell} \left(f\left(\hat{\theta}\right), \theta\right) \end{array} \right\}$$
(5)

For all $\beta \in \mathcal{B}^t$, let $\mathcal{R}(\beta) = \prod_{i \in \mathcal{I}} \mathcal{R}_i(\beta)$.¹¹

To proceed further, we present some useful properties of \mathcal{R} . To this end, we need additional notation. Let $\mathcal{E}(\mathcal{R})$ denote the set of fixed points of \mathcal{R} , which is defined by

$$\mathcal{E}(\mathcal{R}) = \left\{ \beta \in \mathcal{B}^t | \mathcal{R}(\beta) = \beta \right\}$$

The function $\mathcal{R} : \mathcal{B}^t \mapsto \mathcal{B}^t$ is monotone (or, increasing) on \mathcal{B}^t if for all $\beta, \beta' \in \mathcal{B}^t$, $\beta \subseteq \beta'$ implies $\mathcal{R}(\beta) \subseteq \mathcal{R}(\beta')$. For each $\theta \in \Theta$, let us define \mathcal{I}^{θ} by $\mathcal{I}^{\theta} = \{i \in \mathcal{I} | SL_i(f(\theta), \theta) \neq \emptyset\}$. Finally, let $\{\beta^k\}_{k\geq 0}$ be an increasing sequence (in the sense of set inclusion) of elements of \mathcal{B}^t , which is defined recursively as follows. The initial point is the special deception profile β^0 , where $\beta_i^0(\theta)$ is defined in (10) for all $i \in \mathcal{I}$ and all $\theta \in \Theta$, whereas for all $k \geq 1$, β^k is the computation of the mapping \mathcal{R} at the profile β^{k-1} ; that is,

$$\beta^k = \mathcal{R}\left(\beta^{k-1}\right). \tag{6}$$

The first property is that $\beta_i(\theta) \subseteq \mathcal{R}_i^\beta(\theta)$ for all $\beta \in \mathcal{B}^t$, all $\theta \in \Theta$ and all $i \in \mathcal{I}$. This property follows directly from (5) by setting $E_i = E_j = \{\theta\}$ for all $i, j \in \mathcal{I}$ and $\hat{\theta} = \theta' = \theta$. The second property is that \mathcal{R} is monotone on \mathcal{B}^t .¹² The third property is that the set $\mathcal{E}(\mathcal{R})$ is a complete lattice and that the limit point of the sequence $\{\beta^k\}_{k\geq 0}$ is the smallest element of $\mathcal{E}(\mathcal{R})$. This property comes from the fact that since \mathcal{B}^t is a complete lattice and since \mathcal{R} is monotone, Tarski fixed point-theorem implies that \mathcal{R} has a greatest and a least fixed point and that $\mathcal{E}(\mathcal{R})$ is a complete lattice. Moreover, since \mathcal{B}^t is finite, the mapping \mathcal{R} is continuous. Therefore, by applying Echenique (2005), we can conclude that the limit point of the sequence $\{\beta^k\}_{k\geq 0}$, denoted by β^* , is the smallest fixed point of $\mathcal{E}(\mathcal{R})$. The last property says that when if $\bigcap_{j\in\mathcal{I}\setminus\{i\}}\beta_j^*(\theta)\bigcap\Theta_i^f\neq \emptyset$, then $\beta_i^*(\theta)=\Theta^{.13}$

¹¹Alternatively, \mathcal{R}^{β} can be defined via an iterative elimination notion that reflects the notion of iterative elimination of never best-responses. For such an approach, we refer the reader to Jain et al. (2021).

¹²It directly follows from (5) and the assumption that $\beta \subseteq \beta'$.

¹³To see it, let $\theta^* \in \bigcap_{j \in \mathcal{I} \setminus \{i\}} \beta_j^*(\theta) \cap \Theta_i^f$. Then, it follows from (10) that $\beta_i^0(\theta^*) = \Theta$. Since

Since the arguments from the previous paragraph clarify why the mapping \mathcal{R} has the above-discussed properties, we summarize them in the following lemma, whose proof is omitted.

Lemma 1. $\mathcal{R} : \mathcal{B}^t \mapsto \mathcal{B}^t$ has the following properties.

- (*i*) For all $\beta \in \mathcal{B}^t$, $\beta \subseteq \mathcal{R}(\beta)$.
- (*ii*) \mathcal{R} is monotone on \mathcal{B}^t .
- (*iii*) $\mathcal{E}(\mathcal{R})$ is a complete lattice and $\{\beta^*\} = \min \mathcal{E}(\mathcal{R})$ where $\beta^* \equiv \sup_{k>0} \beta^k$.
- (*iv*) For all $i \in \mathcal{I}$ and all $\theta \in \Theta$, if $\bigcap_{j \in \mathcal{I} \setminus \{i\}} \beta_j^*(\theta) \cap \Theta_i^f \neq \emptyset$, then $\beta_i^*(\theta) = \Theta$.

The following condition is at the heart of our characterization result.

Definition 4. $f : \Theta \mapsto Y$ satisfies *Iterative Monotonicity* (henceforth, IM) if for all $\theta, \theta' \in \Theta$,

$$f(\theta) \neq f(\theta') \implies \beta^*(\theta) \bigcap \beta^*(\theta') = \emptyset.$$
 (7)

Let us here briefly explain IM and its relationship with rationalizable implementation. Recall that Maskin monotonicity ensures the elimination of undesirable Nash equilibrium. Similarly, IM ensures the elimination of undesirable 'best-response sets.' This is done by computing players' largest best-response sets via the fixed point deception profile β^* .

To see it, suppose that \mathcal{M} rationalizable implements f, so that $S^{\mathcal{M},\theta}$ is the set of rationalizable strategy profiles for θ . Recall that β^* is the limit point of the sequence $\{\beta^k\}_{k\geq 0}$ and that this sequence is an increasing sequence (in the sense of set inclusion) of elements of \mathcal{B}^t , which is defined recursively. Its starting deception is β^0 , where β_i^0 is defined in (10) for all $i \in \mathcal{I}$ and all $\theta \in \Theta$, whereas for all $k \geq 1$, β^k is computed by $\mathcal{R}(\beta^{k-1})$ as in (6).

 $[\]beta^* \in \mathcal{B}^t$, it follows that $\beta_i^*(\theta^*) = \Theta$. To show that $\beta_i^*(\theta) = \Theta$, it suffices to show that $\Theta \subseteq \mathcal{R}_i^{\beta^*}(\theta)$. This follows from (5) by setting $E_i = \Theta$ and $E_j = \beta_j^*(\theta)$ for all $j \in \mathcal{I} \setminus \{i\}$ and by setting $\tilde{\theta} = \hat{\theta} = \theta^*$ for all $\theta' \in E_i$ and setting $\tilde{\theta} = \hat{\theta} = \theta$ for all $\theta' \in E_j$.

For every $\theta \in \Theta$ and every player *i*, implementability requires that the set $\beta^{0}(\theta)$ generates a best-response set at θ that cannot be eliminated; that is, for each $i \in \mathcal{I}$, $\bigcup_{\bar{\theta}\in\beta_i^0(\theta)}S_i^{\mathcal{M},\bar{\theta}}\subseteq S_i^{\mathcal{M},\theta}.$

By applying (5), it is possible to compute $\mathcal{R}^{\beta^{0}}(\theta) = \beta^{1}(\theta)$. Since, by construction, $\beta^{1}(\theta)$ is computed in a way that it generates a best-response set at θ that cannot be eliminated, it holds, for each $i \in \mathcal{I}$, that $\bigcup_{\bar{\theta} \in \beta_i^1(\theta)} S_i^{\mathcal{M},\bar{\theta}} \subseteq S_i^{\mathcal{M},\theta}$.

This reasoning can be repeated to derived a best-response set at θ generated by $\mathcal{R}^{\beta^{1}}(\theta) = \beta^{2}(\theta)$. And so on. After a finite number of iterations, the largest bestresponse set at θ that cannot be eliminated is obtained by the limit point deception β^* ; that is, for each $i \in \mathcal{I}$, it holds that

$$\bigcup_{\bar{\theta}\in\beta_i^*(\theta)} S_i^{\mathcal{M},\bar{\theta}} \subseteq S_i^{\mathcal{M},\theta}.$$
(8)

From this perspective, β^* can be viewed as the largest deception that is not refutable given the mechanism \mathcal{M} . To see that f satisfies IM if it is rationalizable implementable by the mechanism \mathcal{M} , fix any $\theta, \theta' \in \Theta$ such that $f(\theta) \neq f(\theta')$. Let us show that the intersection $\beta^*(\theta) \cap \beta^*(\theta')$ is empty. To see it, suppose that it is not empty. This implies that for each player *i*, it holds that $\theta(i) \in \beta_i^*(\theta) \cap \beta_i^*(\theta')$. Since \mathcal{M} rationalizable implements f, it follows from (8) that the intersection $S_i^{\mathcal{M},\theta} \cap S_i^{\mathcal{M},\theta'}$ is nonempty for every player *i*, which leads to the contradiction that $f(\theta) \neq f(\theta')$.

The following theorem shows that IM is necessary and sufficient for rationalizable implementation.

Theorem 1. $f: \Theta \mapsto Y$ is rationalizably implementable if and only if f satisfies IM.

For the proof, see Appendices A and B.

V. CONNECTION WITH THE PARTITION BASED APPROACH

In this section, we connect our approach with the partition-based approach followed by BMT and Xiong (2022). For this reason, we restrict our discussion below to the case of three or more players. Recall that the implementing conditions in the existing literature rely on the existence of a partition P of Θ . To make a connection between the two approaches, let \mathcal{B}_f denote the set of deceptions that satisfy (7). Our approach can be summarized as follows: An SCF f is rationalizably implementable if, and only if, the deception $\beta^* \in \mathcal{B}_f$. When f satisfies IM, we can construct a partition $P \in \mathcal{P}_f$ by using β^* as follows:

$$P(\theta) = \left\{ \tilde{\theta} \in \Theta \middle| \beta^*(\theta) = \beta^*(\tilde{\theta}) \right\}$$
(9)

for all $\theta \in \Theta$. If f satisfies IM, it can be checked that P is finer than P_f .¹⁴

Let us first connect our approach with that used by BMT. These authors discuss the role of partition in their characterization result. In particular, they show that the required partition must be as fine as P_f and as coarse as the partition obtained by their Lemma 1, which BMT call "pairwise inclusion property" (See BMT, p. 1266 for a discussion). ¹⁵ BMT show that f is rationalizably implementable if f satisfies the NWA condition and Maskin monotonicity and, moreover, it is responsive.¹⁶ When fsatisfies these conditions, it can be shown that the truth-telling deception, $\beta^* = \beta^0 = \beta^t$.

Xiong (2022) shows that rationalizable implementation of an SCF is equivalent to the Strict Event Monotonicity^{**} (SEM^{**}). We show below that SEM^{**} is equivalent to IM.

¹⁴Formal arguments can be found in Appendix C, where we prove that IM is equivalent to the necessary and sufficient conditions of Xiong (2022).

¹⁵Indeed, BMT argue that the pairwise inclusion property is insufficient to pin down the partition by stating "We finally observe that the partition P may yet have to be coarser than is indicated by the pairwise inclusion property", BMT, p. 1266.

¹⁶ f is responsive provided that for all $\theta, \theta' \in \Theta$ such that $\theta \neq \theta'$, it holds that $f(\theta) \neq f(\theta')$.

Theorem 2. Assume that $I \geq 3$ and that that $\mathcal{I}^{\Theta} \neq \emptyset$. f satisfies IM if and only if it is SEM^{**}.

The proof of the above theorem can be found in Appendix C. Let us briefly discuss it below. When f satisfies IM, the partition used to show that f is SEM^{**} is defined in (9). For the converse result, we show that the partition P appearing in SEM^{**} specifies a deception β^P such that $\mathcal{R}(\beta^P) = \beta^P$ and $\beta^P \in \mathcal{B}_f$. More precisely, β^P is defined by:

$$\beta_i^P(\theta) = \begin{cases} P(\theta) & \text{if } i \in \mathcal{I}^{P(\theta)} \\ \Theta & \text{if } i \notin \mathcal{I}^{P(\theta)}. \end{cases}$$
(10)

where $i \in \mathcal{I}^{P(\theta)}$ provided that $P(\theta) \bigcap \Theta_i^f = \emptyset$; otherwise, $i \notin \mathcal{I}^{P(\theta)}$. When f is SEM^{**}, the limit deception β^* of our iterative procedure is explicitly defined by (10).

In summary, the fundamental novelty of our approach is to uncover the mapping \mathcal{R} , which is implicitly used in the existing characterizations. This mapping plays a critical role in the analysis of two-player implementation problems. Indeed, when we focus on partitions in \mathcal{P}_f , we implicitly restrict the set of deceptions in \mathcal{B}_f to the set $\mathcal{B}_f(\mathcal{P}_f) = \left\{ \beta \in \mathcal{B}_f \middle| \beta = \beta^P \text{ for some } P \in \mathcal{P}_f \right\}$. When there are three or more players, this restriction is without loss of generality. The reason is that for any partition $P \in \mathcal{P}_f$ such that $\beta^P \in \mathcal{E}(\mathcal{R})$, implementation requires that $\beta^0 \subseteq \beta^P$. It is clear from the definition of β^0 that this constraint is vacuously satisfied, when there are three or more players.¹⁷ However, this is not the case for studying two-player implementation problems. Indeed, in what follows, we present an example in which there does not exist any partition $P \in \mathcal{P}_f$ such that $\beta^0 \subseteq \beta^P \in \mathcal{E}(\mathcal{R})$, though f is rationalizably implementable (that is, f satisfies IM with respect to $\beta^* = \beta^0$).

The example has two players, denoted by 1 and 2, three states, denoted by θ , θ' and θ'' , and six outcomes, denoted by a, b, c, d, e and f. Players' utilities from pure

¹⁷Indeed, $\beta^0 = \beta \underline{P}$.

outcomes are summarized in the table below, where $\varepsilon \in (\frac{1}{2}, 1)$.

	$u_{1}\left(\cdot, heta ight)$	$u_{2}\left(\cdot, heta ight)$	$u_1\left(\cdot, \theta'\right)$	$u_{2}\left(\cdot,\theta'\right)$	$u_1\left(\cdot, \theta''\right)$	$u_2\left(\cdot, \theta''\right)$
a	1	$-(1-\varepsilon)$	1	-1	1	$-(1-\varepsilon)$
b	0	0	0	0	0	0
c	-1	1	$-(1-\varepsilon)$	1	$-(1-\varepsilon)$	1
d	1	-2	-2	-1	-2	-2
e	2	-3	-2	-2	-2	-3
f	-3	-3	-3	3	-3	-3

The planner wants to implement f, which is defined by

$$f(\theta) = f(\theta') = \{b\}$$
 and $f(\theta'') = \{a\}$.

It can be easily checked that f is NWA. Let us note that P_f is such that $P_f(\theta) =$ $P_f(\theta') = \{\theta, \theta'\}$ and $P_f(\theta'') = \{\theta''\}$. Also, the finest partition \underline{P} is $\underline{P}(\theta) = \{\theta\}$, $\underline{P}(\theta') = \{\theta'\}$ and $\underline{P}(\theta'') = \{\theta''\}$. It can be checked that $\mathcal{P}_f = \{P_f, \underline{P}\}$. In Appendix ??, we show that f is rationalizably implementable.

Finally, let us show that there does not exist any partition $P \in \mathcal{P}_f$ such that $\beta^0 \subseteq \beta^P \in \mathcal{E}(\mathcal{R})$. To this end, we need to consider only the partition β^{P_f} .¹⁸ Observe that, by construction, $\beta^0 \subseteq \beta_f^P$. We show that $\beta^{P_f} \notin \mathcal{E}(\mathcal{R})$. Assume, to the contrary, that $\beta^{P_f} \in \mathcal{E}(\mathcal{R})$. This implies that for each $i \in \mathcal{I}$, it holds that $P_f(\theta) \cap \mathcal{R}_i^{\beta_f^P}(\theta'') = \emptyset$.

Note that, by construction, $u_1(\cdot, \theta') = u_1(\cdot, \theta'')$ and $u_2(\cdot, \theta) = u_2(\cdot, \theta'')$. This implies that $SL_1(f(\theta), \theta') \subseteq L_1(f(\theta), \theta'')$ and $SL_2(f(\theta), \theta) \subseteq L_2(f(\theta), \theta'')$. Thus, in our environment, it is equivalent to $L_1(f(\theta), \theta') \subseteq L_1(f(\theta), \theta'')$ and $\theta' \in P_f(\theta)$ and to $L_{2}(f(\theta), \theta) \subseteq L_{2}(f(\theta), \theta'')$ and $\theta \in P_{f}(\theta)$. Applying (5) under the specification that $E_1 = E_2 = P_f(\theta)$, we can conclude that for each $i \in \mathcal{I}$, it holds that $P_f(\theta) \subseteq \mathcal{R}_i^{\beta_f^P}(\theta'')$, yielding a contradiction.¹⁹

¹⁸The reason is that, by construction, $\beta_1^0(\theta) = \{\theta, \theta'\}$ but $\beta_1^{\underline{P}} = \{\theta\}$. ¹⁹To see it, we need to set $\tilde{\theta} = \hat{\theta} = \theta'$ for player 1 and $\tilde{\theta} = \hat{\theta} = \theta$ for player 2, and observe that

APPENDICES

A. PROOF OF "ONLY IF" PART OF THEOREM 1

Suppose that \mathcal{M} implements f in rationalizable strategies. To save writing, we show below that f satisfies IM when I = 2. The proof for the case $I \geq 3$ is available upon request. Thus, let us assume that I = 2. To show that f satisfies IM, we need the following useful results and notation.

For all $i \in \mathcal{I}$, let λ_i be any player *i*'s belief. The support of λ_i is defined by $supp(\lambda_i) = \{m_{-i} \in M_{-i} | \lambda_i(m_{-i}) > 0\}.$

Lemma 2. For all $i \in \mathcal{I}$ and all $\theta \in \Theta$, there exists $\lambda_i^{\theta} \in \Delta\left(S_{-i}^{\mathcal{M},\theta}\right)$ such that for all $m_i \in S_i^{\mathcal{M},\theta}$, m_i is a best-response to λ_i^{θ} at θ .

Proof. Take any $\theta \in \Theta$ and any $i \in \mathcal{I}$. Since f is rationalizably implementable by \mathcal{M} , it follows that $S^{\mathcal{M},\theta} \neq \emptyset$ and $f(\theta) = g(m)$ for all $m \in S^{\mathcal{M},\theta}$.

Fix any $m_i \in S_i^{\mathcal{M},\theta}$. Then, m_i is a best-response to some $\lambda_i^{m_i,\theta} \in \Delta\left(S_{-i}^{\mathcal{M},\theta}\right)$ at θ . Let $\lambda_i^{m_i,\theta} = \lambda_i^{\theta}$. Fix any $m_i^* \in S_i^{\mathcal{M},\theta}$. Since f is rationalizably implementable by \mathcal{M} , we have that

$$u_{i}(f(\theta),\theta) = \sum_{\substack{m_{-i} \in M_{-i}}} \lambda_{i}^{\theta}(m_{-i}) u_{i}(g(m_{i}^{*},m_{-i}),\theta)$$

$$\geq \sum_{\substack{m_{-i} \in M_{-i}}} \lambda_{i}^{\theta}(m_{-i}) u_{i}(g(m_{i}',m_{-i}),\theta)$$

for all $m'_i \in M_i$. Thus, m^*_i is a best-response to λ^{θ}_i at θ . Since the choice of $m^*_i \in S^{\mathcal{M},\theta}_i$ is arbitrary, the statement follows.

To show that f satisfies IM, we show that (7) is satisfied with respect to $\hat{\beta}$, which is define below, and then we show that $\beta^* \subseteq \hat{\beta}$.

For all $i \in \mathcal{I}$, let $\hat{\beta}_i$ be defined, for all $\theta \in \Theta$, by

 $\overline{\beta_i^{P_f}(\theta) = \beta_i^{P_f}(\theta') = \{\theta, \theta'\} \text{ for all } i \in \mathcal{I}.}$

$$\hat{\beta}_{i}(\theta) = \left\{ \theta' \in \Theta \middle| supp(\lambda_{-i}^{\theta'}) \subseteq S_{i}^{\mathcal{M},\theta} \right\}.$$
(11)

Let us first show that $\hat{\beta} \in \mathcal{B}^t$. Specifically, by (10), we need to show that $\beta_i^0(\theta) \subseteq \hat{\beta}_i(\theta)$ for all $i \in \mathcal{I}$ and all $\theta \in \Theta$. To this end, fix any $i \in \mathcal{I}$, any $\theta \in \Theta$ and any $\theta' \in \beta_i^0(\theta) = A_i(\theta)$. Let us show that $\theta' \in \hat{\beta}_i(\theta)$. Since $\theta \in \hat{\beta}_i(\theta)$, by (11), let us suppose that $\theta' \neq \theta$. Since \mathcal{M} implements f in rationalizable strategies, it follows that

$$g\left[supp\left(\lambda_{-i}^{\theta'}\right) \times supp\left(\lambda_{i}^{\theta}\right)\right] \subseteq L_{i}\left(f\left(\theta\right),\theta\right) \bigcap L_{-i}\left(f\left(\theta'\right),\theta'\right).$$
(12)

Fix any $m_i^* \in supp(\lambda_{-i}^{\theta'})$. Assume, to the contrary, that m_i^* is not a best-response to λ_i^{θ} at θ . Since $m_i^* \in supp(\lambda_{-i}^{\theta'})$ but it is not a best-response to λ_i^{θ} at θ , it follows from (12) that

$$g\left[m_{i}^{*}\times supp\left(\lambda_{i}^{\theta}\right)\right]\subseteq SL_{i}\left(f\left(\theta\right),\theta\right)\bigcap L_{-i}\left(f\left(\theta'\right),\theta'\right).$$

(2) implies that $\theta' \notin A_i(\theta)$, which is a contradiction. Thus, we have established that every element of $supp(\lambda_{-i}^{\theta'})$ is a best-response to λ_i^{θ} at θ if $\theta' \in \beta_i^0(\theta) = A_i(\theta)$. Thus, $supp(\lambda_{-i}^{\theta'}) \subseteq S_i^{\mathcal{M},\theta}$, and so $\theta' \in \hat{\beta}_i(\theta)$, by (11).

Let us now show that (7) is satisfied with respect to $\hat{\beta}$. To this end, take any $\theta, \theta' \in \Theta$ such that $f(\theta) \neq f(\theta')$. We show that $\hat{\beta}(\theta) \cap \hat{\beta}(\theta') = \emptyset$. Assume, to the contrary, that $\hat{\beta}(\theta) \cap \hat{\beta}(\theta') \neq \emptyset$. This implies that for all $i \in \mathcal{I}$, there exists $\theta_i \in \Theta$ such that $\theta_i \in \hat{\beta}_i(\theta) \cap \hat{\beta}_i(\theta')$. Since $\theta_i \in \hat{\beta}_i(\theta) \cap \hat{\beta}_i(\theta')$ for all $i \in \mathcal{I}$, it follows from (11) that $S_i^{\mathcal{M},\theta} \cap S_i^{\mathcal{M},\theta'} \neq \emptyset$ for all $i \in \mathcal{I}$, so that $S^{\mathcal{M},\theta} \cap S^{\mathcal{M},\theta'} \neq \emptyset$. Since \mathcal{M} implements f in rationalizable strategies, we have that $f(\theta) = f(\theta')$, which is a contradiction.

Next, let us show that $\beta^* \subseteq \hat{\beta}$. To this end, we first show that $\mathcal{R}\left(\hat{\beta}\right) = \hat{\beta}$. The following lemmata establishes that $\mathcal{R}\left(\hat{\beta}\right) \subseteq \hat{\beta}$.

Lemma 3. For all $i \in \mathcal{I}$, all $\theta, \theta', \tilde{\theta} \in \Theta$, all $\theta_{-i} \in \Theta_{-i}$ and all $\beta \in \mathcal{B}^t$, if $\beta \subseteq \hat{\beta}$,

 $\theta_{-i} \in \beta_{-i}\left(\tilde{\theta}\right)$, and either

$$\begin{bmatrix} \theta' \in \beta_i \left(\tilde{\theta} \right), L_i \left(f \left(\theta_{-i} \right), \theta_{-i} \right) \subseteq L_i \left(f \left(\tilde{\theta} \right), \theta \right) \\ \text{and} \, \tilde{\theta} \notin \Theta_i^f \end{bmatrix}$$
(13)

or

$$\tilde{\theta} \in \Theta_i^f,$$

then every element of $supp\left(\lambda_{-i}^{\theta'}\right)$ is a best-response to $\lambda_{i}^{\theta_{-i}}$ at θ .

Proof. Fix any $i \in \mathcal{I}$, any $\theta, \theta', \tilde{\theta} \in \Theta$, any $\theta_{-i} \in \Theta_{-i}$ and any $\beta \in \mathcal{B}^t$ such that $\beta \subseteq \hat{\beta}$ and $\theta_{-i} \in \beta_{-i}\left(\tilde{\theta}\right)$. Since $\theta_{-i} \in \beta_{-i}\left(\tilde{\theta}\right)$ and since $\beta\left(\tilde{\theta}\right) \subseteq \hat{\beta}\left(\tilde{\theta}\right)$, it follows from (11) that $supp\left(\lambda_i^{\theta_{-i}}\right) \subseteq S_{-i}^{\mathcal{M},\tilde{\theta}}$. We proceed according to whether (13) holds or $\tilde{\theta} \in \Theta_i^f$.

Case 1: (13) holds

Since $\theta' \in \beta_i\left(\tilde{\theta}\right)$ and since $\beta_i\left(\tilde{\theta}\right) \subseteq \hat{\beta}_i\left(\tilde{\theta}\right)$, it follows from (11) that $supp\left(\lambda_{-i}^{\theta'}\right) \subseteq S_i^{\mathcal{M},\tilde{\theta}}$. Therefore, we have that $supp\left(\lambda_{-i}^{\theta'}\right) \times supp\left(\lambda_i^{\theta_{-i}}\right) \subseteq S^{\mathcal{M},\tilde{\theta}}$. Since \mathcal{M} implements f, it follows that

$$g\left[supp\left(\lambda_{-i}^{\theta'}\right) \times supp\left(\lambda_{i}^{\theta_{-i}}\right)\right] = f\left(\tilde{\theta}\right).$$
(14)

Finally, fix any $m_i^* \in supp(\lambda_{-i}^{\theta'})$. Assume, to the contrary, that m_i^* is not a bestresponse to $\lambda_i^{\theta_{-i}}$ at θ . It follows that there exists $\tilde{m}_i \in M_i$ such that

$$\sum_{m_{-i} \in M_{-i}} \lambda_{i}^{\theta_{-i}}(m_{-i}) u_{i}\left(g\left(\tilde{m}_{i}, m_{-i}\right), \theta\right) > \sum_{m_{-i} \in M_{-i}} \lambda_{i}^{\theta_{-i}}(m_{-i}) u_{i}\left(g\left(m_{i}^{*}, m_{-i}\right), \theta\right).$$

Since $g(m_i^*, m_{-i}) = f(\tilde{\theta})$ for all $m_{-i} \in supp(\lambda_i^{\theta_{-i}})$, by (14), it follows that

$$\sum_{m_{-i}\in M_{-i}}\lambda_{i}^{\theta_{-i}}\left(m_{-i}\right)u_{i}\left(g\left(\tilde{m}_{i},m_{-i}\right),\theta\right)>u_{i}\left(f\left(\tilde{\theta}\right),\theta\right).$$
(15)

The lottery
$$\left(\sum_{m_{-i}\in M_{-i}}\lambda_{i}^{\theta_{-i}}\left(m_{-i}\right)g\left(\tilde{m}_{i},m_{-i}\right)\right)\notin L_{i}\left(f\left(\theta_{-i}\right),\theta_{-i}\right)$$
 because $L_{i}\left(f\left(\theta_{-i}\right),\theta_{-i}\right)\subseteq$

 $L_i\left(f\left(\tilde{\theta}\right), \theta\right)$, by our initial supposition that (13) holds. However, since \mathcal{M} implements f, Lemma 2 implies that it must be the case that $\left(\sum_{m_{-i}\in M_{-i}}\lambda_i^{\theta_{-i}}\left(m_{-i}\right)g\left(\tilde{m}_i,m_{-i}\right)\right) \in L_i\left(f\left(\theta_{-i}\right), \theta_{-i}\right)$, which is a contradiction.

Case 2: $\tilde{\theta} \in \Theta_i^f$

Then, $S_i^{\mathcal{M},\tilde{\theta}} = M_i$. Recall that $supp\left(\lambda_i^{\theta_{-i}}\right) \subseteq S_{-i}^{\mathcal{M},\tilde{\theta}}$. Since \mathcal{M} implements f, it follows that

$$g\left[M_i \times supp\left(\lambda_i^{\theta_{-i}}\right)\right] = f\left(\tilde{\theta}\right).$$

Therefore, every $m_i \in supp\left(\lambda_{-i}^{\theta'}\right)$ is a best-response to $\lambda_i^{\theta_{-i}}$ at θ .

To proceed further, we need additional notation. For all $i \in \mathcal{I}$, all $\theta \in \Theta$ and all $\beta \in \mathcal{B}^t$, let us define the set $\Psi_i^\beta(\mathcal{M}, \theta)$ by

$$\Psi_{i}^{\beta}(\mathcal{M},\theta) = \bigcup_{\theta' \in \mathcal{R}_{i}^{\beta}(\theta)} supp(\lambda_{-i}^{\theta'}).$$
(16)

Lemma 4. For all $\beta \in \mathcal{B}^t$ and all $\theta \in \Theta$, if $\beta \subseteq \hat{\beta}$, then $\Psi^{\beta}(\mathcal{M}, \theta) \subseteq S^{\mathcal{M}, \theta}$.

Proof. Take any $\theta \in \Theta$ and any $\beta \in \mathcal{B}^t$ such that $\beta \subseteq \hat{\beta}$. To show that $\Psi^{\beta}(\mathcal{M}, \theta) \subseteq S^{\mathcal{M},\theta}$, it suffices to show that the set $\Psi^{\beta}(\mathcal{M}, \theta)$ satisfies the best-response property at θ ; that is, it suffices to show that for all $i \in \mathcal{I}$ and all $m_i^* \in \Psi_i^{\beta}(\mathcal{M}, \theta)$, there exists $\lambda_i^{m_i^*} \in \Delta(M_{-i})$ such that $\lambda_i^{m_i^*}(m_{-i}) > 0 \implies m_{-i} \in \Psi_{-i}^{\beta}(\mathcal{M}, \theta)$, and m_i^* is a best-response to $\lambda_i^{m_i^*}$ at θ .

Fix any $i \in \mathcal{I}$ and any $m_i^* \in \Psi_i^{\beta}(\mathcal{M}, \theta)$. Since $m_i^* \in \Psi_i^{\beta}(\mathcal{M}, \theta)$, (16) implies that there exists $E_i \in 2^{\Theta} \setminus \{\emptyset\}$ such that $\theta' \in E_i \subseteq \mathcal{R}_i^{\beta}(\theta)$. (5) implies that there exists $E_{-i} \in 2^{\Theta} \setminus \{\emptyset\}$ such that for each $\ell \in \mathcal{I}$ and each $\theta_\ell \in E_\ell$, there exist $\tilde{\theta} \in E_\ell$ and $\theta_{-\ell} \in \beta_{-\ell}(\tilde{\theta}) \cap E_{-\ell}$ such that $\theta_\ell \in \beta_\ell(\tilde{\theta})$ and either $\tilde{\theta} \in \Theta_\ell^f$ or $L_\ell(f(\theta_{-\ell}), \theta_{-\ell}) \subseteq$ $L_\ell(f(\tilde{\theta}), \theta)$. By (5), it also follows that $E_{-i} \subseteq \mathcal{R}_{-i}^{\beta}(\theta)$, and so $\theta_{-i} \in \mathcal{R}_{-i}^{\beta}(\theta)$. Since $\theta_{-i} \in \mathcal{R}_{-i}^{\beta}(\theta)$, (16) implies that $supp(\lambda_i^{\theta_{-i}}) \subseteq \Psi_{-i}^{\beta}(\mathcal{M}, \theta)$. Moreover, since $\beta \subseteq \hat{\beta}$ and $\theta_{-i} \in \beta_{-i}\left(\tilde{\theta}\right)$, and since, moreover, either $\tilde{\theta} \in \Theta_i^f$ or

$$\theta' \in \beta_i\left(\tilde{\theta}\right), L_i\left(f\left(\theta_{-i}\right), \theta_{-i}\right) \subseteq L_i\left(f\left(\tilde{\theta}\right), \theta\right) \text{ and } \tilde{\theta} \notin \Theta_i^f,$$

it follows from Lemma 3 that every element of $supp(\lambda_{-i}^{\theta'})$ is a best-response to $\lambda_i^{\theta_{-i}}$ at θ . Thus, m_i^* is a best-response to $\lambda_i^{\theta_{-i}}$ at θ and $supp(\lambda_i^{\theta_{-i}}) \subseteq \Psi_{-i}^{\beta}(\mathcal{M}, \theta)$, as we sought.

Lemma 5. For all $\beta \in \mathcal{B}^t$, if $\beta \subseteq \hat{\beta}$, then $\mathcal{R}(\beta) \subseteq \hat{\beta}$.

Proof. Fix any $\beta \in \mathcal{B}^t$ such that $\beta \subseteq \hat{\beta}$. Fix any $i \in \mathcal{I}$ and any $\theta, \theta' \in \Theta$ such that $\theta' \in \mathcal{R}_i^{\beta}(\theta)$. We show that $\theta' \in \hat{\beta}_i(\theta)$. (16) implies that $supp(\lambda_{-i}^{\theta'}) \subseteq \Psi_i^{\beta}(\mathcal{M}, \theta)$. Since $\beta \subseteq \hat{\beta}$, Lemma 4 implies that $\Psi_i^{\beta}(\mathcal{M}, \theta) \subseteq S_i^{\mathcal{M}, \theta}$. Thus, $supp(\lambda_{-i}^{\theta'}) \subseteq S_i^{\mathcal{M}, \theta}$. By definition of $\hat{\beta}$, in (11), it follows that $\theta' \in \hat{\beta}_i(\theta)$, as we sought.

Since $\hat{\beta} \subseteq \hat{\beta}$, Lemma 5 implies that $\mathcal{R}(\hat{\beta}) \subseteq \hat{\beta}$. Since part (i) of Lemma 1 implies that $\mathcal{R}(\hat{\beta}) \supseteq \hat{\beta}$, it follows that $\mathcal{R}(\hat{\beta}) = \hat{\beta}$, and so $\hat{\beta} \in \mathcal{E}(\mathcal{R})$. Since $\{\beta^*\} = \min \mathcal{E}(\mathcal{R})$, by part (iii) of Lemma 1, we have that $\beta^* \subseteq \hat{\beta}$, as we sought. Thus, f satisfies IM.

B. PROOF OF "IF" PART OF THEOREM 1

Suppose that $f: \Theta \to Y$ satisfies IM. Below we provide the proof for I = 2. The reason is that the case $I \ge 3$ can be proved similarly. The details are available from authors. However, indirect arguments can be found in Section ??. Therefore, in what follows, we suppose that I = 2.

For each $i \in \mathcal{I}$, recall the definition of Θ_i^f in (1). The complement of Θ_i^f is denoted by $\bar{\Theta}_i^f$. For each $i \in \mathcal{I}$, $y_i : \bar{\Theta}_i^f \to Y$ is a function such that for each $\theta \in \bar{\Theta}_i^f$,

$$y_i(\theta) \in SL_i(f(\theta), \theta).$$
 (17)

Given the set $\{y_i(\theta)\}_{\theta\in\bar{\Theta}_i^f}$, we define the average lotteries by setting

$$\underline{y}_{i} = \frac{1}{\left|\bar{\Theta}_{i}^{f}\right|} \sum_{\theta \in \bar{\Theta}_{i}^{f}} y_{i}\left(\theta\right) \text{ and } \underline{y} = \frac{1}{I} \sum_{i \in \mathcal{I}} \underline{y}_{i}.$$
(18)

For each $i \in \mathcal{I}, \, y_i^* : \bar{\Theta}_i^f \to Y$ is a function defined, for all $\theta \in \Theta$ by

$$y_i^*(\theta) \in \arg\max_{y \in Y} u_i(y, \theta).$$
 (19)

For each $\theta \in \Theta$, let us define \mathcal{I}^{θ} by $\mathcal{I}^{\theta} = \{i \in \mathcal{I} | SL_i(f(\theta), \theta) \neq \emptyset\}$ and \mathcal{I}^{Θ} by $\mathcal{I}^{\Theta} = \bigcap_{\theta \in \Theta} \mathcal{I}^{\theta}$. To save notation, we omit below the proof for the trivial case $\mathcal{I}^{\Theta} = \emptyset$.²⁰

The following results will be used in constructing \mathcal{M} .

Lemma 6. For all $i \in \mathcal{I}$ and all $\theta \in \overline{\Theta}_i^f$, $u_i(y_i^*(\theta), \theta) > u_i(\underline{y}, \theta)$.

Proof. Formal arguments can be found in Bergemann et al (2011; Lemma 2, p. 1260).

Lemma 7. For each $i \in \mathcal{I}$, there exists a function $z_i : \Theta \times \Theta \to Y$ such that for each $(\theta, \theta') \in \bar{\Theta}_i^f \times \bar{\Theta}_i^f$,

$$u_i(f(\theta'), \theta') > u_i(z_i(\theta, \theta'), \theta'), \qquad (20)$$

and for $\theta \neq \theta'$,

$$u_i(z_i(\theta, \theta'), \theta) > u_i(z_i(\theta', \theta'), \theta).$$

$$(21)$$

Proof. Formal arguments can be found in Bergemann et al (2011; Lemma 2, p. 1260).

Whereas the above results will be used in the construction of both Rule 3 and Rule 4, the following result will be used in the construction of Rule 2.

²⁰The reason is that f is a constant function when $\mathcal{I}^{\Theta} = \emptyset$. See Appendix A of Jain et al. (2021).

Lemma 8. For all $\theta, \theta', \theta'' \in \Theta$, if $(\theta', \theta'') \notin \beta_i^*(\bar{\theta}) \times \beta_{-i}^*(\bar{\theta})$ for all $\bar{\theta} \in \Theta$, then:

(i) There exists $e(\theta', \theta'') \in Y$ such that

$$e\left(\theta',\theta''\right) \in SL_{i}\left(f\left(\theta''\right),\theta''\right) \cap SL_{-i}\left(f\left(\theta'\right),\theta'\right).$$

$$(22)$$

(ii) For all $i \in \mathcal{I}$, if $\theta \in \overline{\Theta}_i^f$, then there exists $y_i^{\theta}(\theta', \theta'') \in Y$ such that

$$u_i\left(y_i^{\theta}\left(\theta',\theta''\right),\theta\right) > u_i\left(e\left(\theta',\theta''\right),\theta\right)$$
(23)

$$y_i^{\theta}\left(\theta',\theta''\right) \in SL_i\left(f\left(\theta''\right),\theta''\right).$$
(24)

Proof. Fix any $\theta, \theta', \theta'' \in \Theta$. Suppose that $(\theta', \theta'') \notin \beta_i^*(\bar{\theta}) \times \beta_{-i}^*(\bar{\theta})$ for all $\bar{\theta} \in \Theta$. Since $\beta^* \in \mathcal{E}(\mathcal{R})$, we have that $(\theta', \theta'') \notin \mathcal{R}_i^{\beta^*}(\bar{\theta}) \times \mathcal{R}_{-i}^{\beta^*}(\bar{\theta})$ for all $\bar{\theta} \in \Theta$. Since $\theta'' \in \mathcal{R}_{-i}^{\beta^*}(\theta'')$, it holds that $\theta' \notin \mathcal{R}_i^{\beta^*}(\theta'')$. Part (i) of Lemma 1 implies that $\beta_i^*(\theta'') \subseteq \mathcal{R}_i^{\beta^*}(\theta'')$. Since $\theta' \notin \mathcal{R}_i^{\beta^*}(\theta'')$, it follows that $\theta' \notin \beta_i^0(\theta'')$. Since $\beta^* \equiv \sup_{k\geq 0} \beta^k$ and since $\beta_i^0 \subseteq \beta_i^*$, by (6), it follows that $\theta' \notin \beta_i^0(\theta'')$. Since $\beta_i^0(\theta'') = A_i(\theta'')$, by (10), it follows from (2) that $x \in SL_i(f(\theta''), \theta'') \cap L_{-i}(f(\theta'), \theta')$ for some $x \in Y$. Similarly, we have that $\theta'' \notin A_{-i}(\theta')$, and so $y \in L_i(f(\theta''), \theta'') \cap SL_{-i}(f(\theta'), \theta')$ for some $x \in Y$. Hence, there exists a small, but positive, number $p \in (0, 1)$ such that $z = (px + (1-p)y) \in SL_i(f(\theta''), \theta'') \cap SL_{-i}(f(\theta'), \theta')$.

$$e\left(\theta',\theta''\right) = (1-\varepsilon)z + \varepsilon y \tag{25}$$

with \underline{y} as defined in (18). Since Θ is finite, we can find a sufficiently small, but positive, $\varepsilon > 0$ such that for all $\theta', \theta'' \in \Theta$, $e(\theta', \theta'') \in SL_i(f(\theta''), \theta'') \cap SL_{-i}(f(\theta'), \theta')$, which proves (22).

Fix any $i \in \mathcal{I}$. Suppose $\theta \in \overline{\Theta}_i^f$. Define

$$y_i^{\theta}\left(\theta',\theta''\right) = (1-\varepsilon) z + \varepsilon y_i^*\left(\theta\right), \qquad (26)$$

with $y_i^*(\theta)$ as defined in (19).

Since Θ and $\overline{\Theta}_i^f$ are finite, we can find a sufficiently small, but positive, $\varepsilon > 0$ such that for all $\theta', \theta'' \in \Theta$ and all $\theta \in \overline{\Theta}_i^f$, $e(\theta', \theta'') \in SL_i(f(\theta''), \theta'') \cap SL_{-i}(f(\theta'), \theta')$, which establishes (22), and $y_i^{\theta}(\theta', \theta'') \in SL_i(f(\theta''), \theta'')$, which establishes (24). Since the only difference between $e(\theta', \theta'')$ and $y_i^{\theta}(\theta', \theta'')$ is that \underline{y} in (25) is replaced by $y_i^*(\theta)$ in (26), and since Lemma 6 implies that $u_i(y_i^*(\theta), \theta) > u_i(\underline{y}, \theta)$, it follows that (23) is proved for all $\theta, \theta', \theta''$ such that $\theta \in \overline{\Theta}_i^f$.

For all $i \in \mathcal{I}$ and all $\theta, \theta' \in \Theta$, let $D_i(\theta', \theta)$ be defined by

$$D_{i}(\theta',\theta) = \left\{ y \in Y | y \in L_{i}(f(\theta'),\theta') \cap SU_{i}(f(\theta'),\theta) \right\},$$
(27)

where for all $x \in Y$, $SU_i(x, \theta) = \{y \in Y | u_i(x, \theta) < u_i(y, \theta)\}$. For every $i \in \mathcal{I}$ and every $\theta \in \overline{\Theta}_i^f$, let $\alpha_i^{\theta} : \Theta \times \Theta \to Y$ be a function such that for every $\theta', \theta'' \in \Theta$,

$$\alpha_i^{\theta}\left(\theta',\theta''\right) \in L_i\left(f\left(\theta'\right),\theta'\right) \cap SU_i\left(f\left(\theta''\right),\theta\right) \tag{28}$$

if $L_i(f(\theta'), \theta') \cap SU_i(f(\theta''), \theta) \neq \emptyset$.

For every $i \in \mathcal{I}$ and every $\theta, \theta' \in \Theta$, let $\overline{D}_i(\theta', \theta)$ be any finite subset of $D_i(\theta', \theta)$ satisfying the following requirements:

- (i) If $\theta, \theta' \in \overline{\Theta}_i^f$ and $z_i(\theta, \theta') \in D_i(\theta', \theta)$, then $z_i(\theta, \theta') \in \overline{D}_i(\theta', \theta)$.
- (ii) For all $\theta'' \in \Theta$, if $\theta \in \bar{\Theta}_i^f$, $(\theta', \theta'') \notin \beta_i^*(\bar{\theta}) \times \beta_{-i}^*(\bar{\theta})$ for all $\bar{\theta} \in \Theta$ and $y_i^{\theta}(\theta', \theta'') \in D_i(\theta'', \theta)$, then $y_i^{\theta}(\theta', \theta'') \in \bar{D}_i(\theta'', \theta)$.
- (iii) For all $\theta'' \in \Theta$, if $\theta \in \overline{\Theta}_i^f$ and $L_i(f(\theta'), \theta') \cap SU_i(f(\theta''), \theta) \neq \emptyset$, then $\alpha_i^{\theta}(\theta', \theta'') \in \overline{D}_i(\theta', \theta)$.

For every $i \in \mathcal{I}$ and every $\theta, \theta' \in \Theta$, fix any $\overline{D}_i(\theta', \theta)$ satisfying the above require-

ments. Let $B_i(\theta', \theta)$ be defined by

$$B_{i}(\theta',\theta) = \begin{cases} \arg \max_{y \in \bar{D}_{i}(\theta',\theta)} u_{i}(y,\theta) & \text{if } D_{i}(\theta',\theta) \neq \emptyset \\ f(\theta') & \text{otherwise.} \end{cases}$$
(29)

Note that $B_i(\theta', \theta) \neq \emptyset$ for all $\theta, \theta' \in \Theta$ and all $i \in \mathcal{I}$. Furthermore, for every $i \in \mathcal{I}$, let $x_i : \Theta \times \Theta \to Y$ be a function such that

$$x_i\left(\theta',\theta\right) \in B_i\left(\theta',\theta\right) \tag{30}$$

for all $\theta', \theta \in \Theta$.

Lemma 9. For all $i \in \mathcal{I}$ and all $\theta, \theta' \in \Theta$, if $\theta, \theta' \in \overline{\Theta}_i^f$, then $u_i(x_i(\theta', \theta), \theta) > u_i(z_i(\theta', \theta'), \theta)$.

Proof. Fix any $i \in \mathcal{I}$ and any $\theta, \theta' \in \overline{\Theta}_i^f$. Then, $x_i(\theta', \theta) \in B_i(\theta', \theta)$ by (30). Let us proceed according to whether $D_i(\theta', \theta) = \emptyset$ or not.

Suppose that $D_i(\theta', \theta) = \emptyset$, so that $x_i(\theta', \theta) = f(\theta')$ by (29). By (20), $u_i(f(\theta'), \theta') > u_i(z_i(\theta, \theta'), \theta')$. This proves the statement if $\theta = \theta'$. Suppose that $\theta \neq \theta'$. By (21), $u_i(z_i(\theta, \theta'), \theta) > u_i(z_i(\theta', \theta'), \theta)$. Since $D_i(\theta', \theta) = \emptyset$ and since $u_i(f(\theta'), \theta') > u_i(z_i(\theta, \theta'), \theta')$, it holds that $u_i(x_i(\theta', \theta), \theta) \ge u_i(z_i(\theta, \theta'), \theta)$, and so $u_i(x_i(\theta', \theta), \theta) > u_i(z_i(\theta', \theta'), \theta)$.

Suppose that $D_i(\theta',\theta) \neq \emptyset$. Then, $\theta \neq \theta'$. Suppose that $u_i(f(\theta'),\theta) \geq u_i(z_i(\theta,\theta'),\theta)$. By (21), $u_i(f(\theta'),\theta) > u_i(z_i(\theta',\theta'),\theta)$. Since $x_i(\theta',\theta) \in B_i(\theta',\theta)$, it follows that $u_i(x_i(\theta',\theta),\theta) \geq u_i(f(\theta'),\theta)$. Therefore, $u_i(x_i(\theta',\theta),\theta) > u_i(z_i(\theta',\theta'),\theta)$. Finally, suppose that $u_i(z_i(\theta,\theta'),\theta) > u_i(f(\theta'),\theta)$. By (20), $u_i(f(\theta'),\theta') > u_i(z_i(\theta,\theta'),\theta')$. Thus, $z_i(\theta,\theta') \in D_i(\theta',\theta)$, by (27), and so $z_i(\theta,\theta') \in \overline{D}_i(\theta',\theta)$, by requirement (i) of the set $\overline{D}_i(\theta',\theta)$. Since $x_i(\theta',\theta) \in B_i(\theta',\theta)$, it follows that $u_i(x_i(\theta',\theta),\theta) \geq u_i(z_i(\theta,\theta'),\theta)$.

Lemma 10. For all $i \in \mathcal{I}$ and all $\theta, \theta', \theta'' \in \Theta$, if $\theta, \theta'' \in \overline{\Theta}_i^f$ and $(\theta', \theta'') \notin \beta_i^*(\overline{\theta}) \times \beta_{-i}^*(\overline{\theta})$ for all $\overline{\theta} \in \Theta$, then $u_i(x_i(\theta'', \theta), \theta) > u_i(e(\theta', \theta''), \theta)$.

Proof. Fix any $i \in \mathcal{I}$ and any $\theta, \theta', \theta'' \in \Theta$ such that $\theta, \theta'' \in \overline{\Theta}_i^f$. Then, $x_i(\theta'', \theta) \in B_i(\theta'', \theta)$ by (30). Suppose that $(\theta', \theta'') \notin \overline{\Psi}_i(\overline{\theta}) \times \overline{\Psi}_{-i}(\overline{\theta})$ for all $\overline{\theta} \in \Theta$. We proceed according to whether $D_i(\theta'', \theta) = \emptyset$ or not.

Suppose that $D_i(\theta'',\theta) = \emptyset$. Thus, $x_i(\theta'',\theta) = f(\theta'')$ by (29). Since $D_i(\theta'',\theta) = \emptyset$ and since (24) holds, it follows that $u_i(x_i(\theta'',\theta),\theta) \ge u_i(y_i^{\theta}(\theta',\theta''),\theta)$. Since (23) holds, we conclude that $u_i(x_i(\theta'',\theta),\theta) > u_i(e(\theta',\theta''),\theta)$.

Suppose that $D_i(\theta'',\theta) \neq \emptyset$. Then, $\theta \neq \theta''$. Suppose that $u_i(f(\theta''),\theta) \ge u_i(y_i^{\theta}(\theta',\theta''),\theta)$. Since $x_i(\theta'',\theta) \in B_i(\theta'',\theta)$, then $u_i(x_i(\theta'',\theta),\theta) \ge u_i(f(\theta''),\theta)$, and so $u_i(x_i(\theta'',\theta),\theta) \ge u_i(y_i^{\theta}(\theta',\theta''),\theta)$. By (23), we have that $u_i(x_i(\theta'',\theta),\theta) > u_i(e(\theta',\theta''),\theta)$. Otherwise, let $u_i(y_i^{\theta}(\theta',\theta''),\theta) > u_i(f(\theta''),\theta)$. By (24), it holds that $u_i(f(\theta''),\theta'') > u_i(y_i^{\theta}(\theta',\theta''),\theta'')$. Thus, $y_i^{\theta}(\theta',\theta'') \in D_i(\theta',\theta)$, and so $y_i^{\theta}(\theta',\theta'') \in \overline{D}_i(\theta',\theta)$, by requirement (ii) of the set $\overline{D}_i(\theta',\theta)$. Since $x_i(\theta'',\theta) \in B_i(\theta'',\theta)$, it follows that $u_i(x_i(\theta',\theta),\theta) \ge u_i(y_i^{\theta}(\theta',\theta''),\theta)$ and so, by (23), $u_i(x_i(\theta'',\theta),\theta) > u_i(e(\theta',\theta''),\theta)$.

Based on the above results, let us construct $\mathcal{M} = (M, g)$. We define the following countable set of lotteries:

$$\mathcal{Y} = \{ z_i(\theta',\theta) \}_{i \in \mathcal{I}, \theta, \theta' \in \bar{\Theta}_i^f} \bigcup \{ y_i^*(\theta) \}_{i \in \mathcal{I}, \theta \in \bar{\Theta}_i^f} \bigcup \{ B_i(\theta',\theta) \}_{i \in \mathcal{I}, \theta, \theta' \in \Theta} \bigcup \{ e(\theta',\theta''), y_i^{\theta}(\theta',\theta'') \}_{i \in \mathcal{I}, \{(\theta',\theta'') \mid (\theta',\theta'') \notin \cup_{\bar{\theta} \in \Theta} (\beta_i^*(\bar{\theta}) \times \beta_{-i}^*(\bar{\theta})) \}},$$

where the collection $\{z_i(\theta',\theta)\}_{i\in\mathcal{I},\theta,\theta'\in\bar{\Theta}_i^f}$ has been defined in Lemma 7, the collection $\{y_i^*(\theta)\}_{i\in\mathcal{I},\theta\in\bar{\Theta}_i^f}$ follows from (23), the collection

$$\left\{e\left(\theta',\theta''\right),y_{i}^{\theta}\left(\theta',\theta''\right)\right\}_{i\in\mathcal{I},\left\{\left(\theta',\theta''\right)|\left(\theta',\theta''\right)\notin\cup_{\bar{\theta}\in\Theta}\left(\beta_{i}^{*}\left(\bar{\theta}\right)\times\beta_{-i}^{*}\left(\bar{\theta}\right)\right)\right\}}$$

is established by Lemma 8, and the collection $\{B_i(\theta',\theta)\}_{i\in\mathcal{I},\theta,\theta'\in\Theta}$ follows from (29).

Each $i \in \mathcal{I}$ plays a strategy $m_i = (m_i^1, m_i^2, m_i^3, m_i^4)$, where $m_i^1 \in \Theta$, $m_i^2 \in \mathbb{Z}_+$, $m_i^3 : \Theta \to \mathcal{Y}$ and $m_i^4 \in \mathcal{Y}$. By construction, M_i is a nonempty countable set for player *i*. Moreover, the third component of the strategy allows player *i* to announce a lottery in \mathcal{Y} contingent on other player playing $m_{-i}^1 = \theta$. The outcome function also uses \underline{y} , as defined in (18). Recall that for all $\theta \in \Theta$, $\mathcal{I}^{\theta} = \{i \in \mathcal{I} | SL_i(f(\theta), \theta) \neq \emptyset\}$ and let $\mathcal{I}^{\Theta} = \bigcap_{\theta \in \Theta} \mathcal{I}^{\theta}$. Thus, $i \in \mathcal{I}^{\theta}$ if and only if $\theta \in \overline{\Theta}_i^f$, if and only if $\theta \notin \Theta_i^f$. For all $\theta \in \Theta$, let

$$\mathcal{I}^{\beta^{*}(\theta)} = \left\{ i \in \mathcal{I} | \beta_{-i}^{*}(\theta) \subseteq \bar{\Theta}_{i}^{f} \right\}.$$
(31)

Note that if $\mathcal{I}^{\beta^*(\theta)} \neq \emptyset$, then $\theta \in \bar{\Theta}_i^f$ for all $i \in \mathcal{I}^{\beta^*(\theta)}$.

Before defining the outcome function g, let us derive the following useful results.

Lemma 11. If $f : \Theta \mapsto Y$ satisfies IM, then for all $\theta, \theta', \overline{\theta} \in \Theta$ and all $i \in \mathcal{I}$, if $\theta' \in \beta_i^*(\overline{\theta})$ and $\theta \in \beta_{-i}^*(\overline{\theta})$, then $f(\overline{\theta}) \in L_i(f(\theta), \theta)$.

Proof. Let the premises hold. Take any $\theta, \theta', \bar{\theta} \in \Theta$ and any $i \in \mathcal{I}$. Suppose that $\theta' \in \beta_i^*(\bar{\theta})$ and $\theta \in \beta_{-i}^*(\bar{\theta})$. We show that $u_i(f(\theta), \theta) \ge u_i(f(\bar{\theta}), \theta)$. Since $\theta \in \beta_{-i}^*(\bar{\theta})$ and since $\theta \in \beta_{-i}^*(\theta)$, it follows that

$$\theta \in \beta_{-i}^{*}\left(\bar{\theta}\right) \bigcap \beta_{-i}^{*}\left(\theta\right).$$
(32)

We proceed according to whether $\theta \notin \overline{\Theta}_i^f$ or not.

Suppose that $\theta \in \Theta_i^f$. Lemma 1 implies that $\beta_i^*(\theta) = \Theta$. Since $\theta' \in \beta_i^*(\overline{\theta})$, it follows that

$$\theta' \in \beta_i^* \left(\bar{\theta} \right) \bigcap \beta_i^* \left(\theta \right). \tag{33}$$

Since (32) and (33) hold, IM implies that $f(\theta) = f(\bar{\theta})$.

Suppose that $\theta \in \bar{\Theta}_{i}^{f}$. Since $u_{i}\left(f\left(\bar{\theta}\right), \theta\right) > u_{i}\left(f\left(\theta\right), \theta\right)$, it follows that $L_{i}\left(f\left(\theta\right), \theta\right) \subseteq L_{i}\left(f\left(\bar{\theta}\right), \theta\right)$. Let us show that $\beta_{i}^{*}\left(\bar{\theta}\right) \subseteq \beta_{i}^{*}\left(\theta\right)$. Since $\beta^{*} = \mathcal{R}\left(\beta^{*}\right)$, it suffices to show that $\beta_{i}^{*}\left(\bar{\theta}\right) \subseteq \mathcal{R}_{i}^{\beta^{*}}\left(\theta\right)$. To this end, let $E_{i} = \left\{\theta, \bar{\theta}\right\}$ and $E_{-i} = \left\{\theta\right\}$. It suffices to show that $\beta_{i}^{*}(E_{i}) \subseteq \mathcal{R}_{i}^{\beta^{*}}\left(\theta\right)$. Let us first consider player -i. Then, for all $\theta'' \in \beta_{-i}^{*}(E_{-i})$, it holds that $\theta \in \beta_{-i}^{*}(E_{-i}), \ \theta \in \beta_{i}^{*}\left(\theta\right) \cap \beta_{i}^{*}(E_{i}), \ \theta'' \in \beta_{-i}^{*}\left(\theta\right)$ and $L_{-i}\left(f\left(\theta\right), \theta\right) \subseteq L_{-i}\left(f\left(\theta\right), \theta\right)$. Next, let us consider player i. Fix any $\theta'' \in \beta_{i}^{*}(E_{i})$.

Let us proceed according to whether $\theta'' \in \beta_i^*(\bar{\theta})$ or not.

- Suppose that $\theta'' \in \beta_i^*(\bar{\theta})$. By our initial supposition, it holds that $\theta \in \beta_{-i}^*(\bar{\theta})$. Moreover, we are also under the assumption that $L_i(f(\theta), \theta) \subseteq L_i(f(\bar{\theta}), \theta)$. Then, we have that $\bar{\theta} \in \beta_i^*(E_i), \ \theta \in \beta_{-i}^*(\bar{\theta}) \cap \beta_{-i}^*(E_{-i}), \ \theta'' \in \beta_i^*(\bar{\theta})$ and $L_i(f(\theta), \theta) \subseteq L_i(f(\bar{\theta}), \theta)$.
- Suppose that $\theta'' \notin \beta_i^*(\bar{\theta})$. Then, $\theta'' \in \beta_i^*(\theta)$. Then, it holds that $\theta \in \beta_i^*(E_i)$, $\theta'' \in \beta_i^*(\theta), \theta \in \beta_{-i}^*(\theta) \cap \beta_{-i}^*(E_{-i})$ and $L_i(f(\theta), \theta) \subseteq L_i(f(\theta), \theta)$.

(5) implies that $\beta_i^*(E_i) \subseteq \mathcal{R}_i^{\beta^*}(\theta)$. Thus, $\beta_i^*(\bar{\theta}) \subseteq \mathcal{R}_i^{\beta^*}(\theta)$.

Since $\theta' \in \beta_i^*(\bar{\theta})$ and $\beta_i^*(\bar{\theta}) \subseteq \mathcal{R}_i^{\beta^*}(\theta) = \beta_i^*(\theta)$, we have that (33) holds. Again, since (32) and (33) hold, IM implies $f(\theta) = f(\bar{\theta})$, which is a contradiction.

For all $m \in M$, the outcome g(m) is defined by the following rules.²¹

Rule 1: If there exists $\bar{\theta} \in \Theta$ such that $(m_i^1)_{i \in \mathcal{I}} \in \beta^* (\bar{\theta})$ and $m_i^2 = 0$ for all $i \in \mathcal{I}^{\beta^*(\bar{\theta})}$, then

$$g(m) = f(\overline{\theta})$$
.

Rule 2: If $m_1^2 = m_2^2 = 0$ and $(m_i^1)_{i \in \mathcal{I}} \notin \beta^* (\bar{\theta})$ for all $\bar{\theta} \in \Theta$, then

$$g\left(m\right) = e\left(m_{i}^{1}, m_{-i}^{1}\right),$$

²¹To include the case $\mathcal{I}^{\Theta} = \emptyset$, Rule 1 needs to be formulated as follows: If there exists $\bar{\theta} \in \Theta$ such that $(m_i^1)_{i \in \mathcal{I}} \in \beta^*(\bar{\theta})$ and $[m_i^2 = 0$ for all $i \in \mathcal{I}^{\beta^*(\bar{\theta})}$ or $\mathcal{I}^{\Theta} = \emptyset$], then $g(m) = f(\bar{\theta})$. It is easy to see that any $m \in M$ falls into Rule 1 when $\mathcal{I}^{\Theta} = \emptyset$.

where the existence of $e(m_i^1, m_{-i}^1)$ is established in Lemma 8.

Rule 3: For all $i \in \mathcal{I}$, if for some $\bar{\theta} \in \bar{\Theta}_i^f$, $(m_{-i}^1, m_{-i}^2) = (\bar{\theta}, 0)$ and $m_i^2 > 0$, then

$$g(m) = \begin{cases} \left(\frac{m_i^2}{m_i^2+1}\right) m_i^3\left(\bar{\theta}\right) + \left(\frac{1}{m_i^2+1}\right) z_i\left(\bar{\theta},\bar{\theta}\right) & \text{if } u_i\left(f\left(\bar{\theta}\right),\bar{\theta}\right) \ge u_i\left(m_i^3\left(\bar{\theta}\right),\bar{\theta}\right) \\ z_i\left(\bar{\theta},\bar{\theta}\right) & \text{otherwise,} \end{cases}$$

where the existence of $z_i(\bar{\theta}, \bar{\theta})$ is established in Lemma 7.

Rule 4: In all other cases, an integer game is played: we identify a pivotal player *i* by requiring that $m_i^2 \ge m_{-i}^2$, and that if $m_i^2 = m_{-i}^2$, then i < -i. Then,

$$g\left(m\right) = \left(\frac{m_i^2}{m_i^2 + 1}\right)m_i^4 + \left(\frac{1}{m_i^2 + 1}\right)\underline{y},$$

where y is defined in (18).

To check that g is well-defined, we need only to check that Rule 1 is well-defined. Assume, to the contrary, that there exists $m \in M$ falling into Rule 1 such that for some $\bar{\theta}, \bar{\theta}' \in \Theta$, $(m_i^1)_{i \in \mathcal{I}} \in \beta^*(\bar{\theta})$ and $m_i^2 = 0$ for all $i \in \mathcal{I}^{\beta^*(\bar{\theta})}$, $(m_i^1)_{i \in \mathcal{I}} \in \beta^*(\bar{\theta}')$ and $m_i^2 = 0$ for all $i \in \mathcal{I}^{\beta^*(\bar{\theta}')}$, and $f(\bar{\theta}) \neq f(\bar{\theta}')$. Then, $m_i^1 \in \beta_i^*(\bar{\theta}) \cap \beta_i^*(\bar{\theta}')$ for all $i \in \mathcal{I}$, and so $\beta^*(\bar{\theta}) \cap \beta^*(\bar{\theta}') \neq \emptyset$. Since f satisfies IM, we have that $f(\bar{\theta}) = f(\bar{\theta}')$, which is a contradiction.

Let $\bar{m}_i = (\theta, 0, m_i^3, m_i^4) \in M_i$, for all $i \in \mathcal{I}$. Let us first show that $\bar{m} \in NE(\mathcal{M}, \theta)$, that is, for all $i \in \mathcal{I}$,

$$u_i\left(g\left(\bar{m}_i, \bar{m}_{-i}\right), \theta\right) \ge u_i\left(g\left(m_i, \bar{m}_{-i}\right), \theta\right),\tag{34}$$

for all $m_i \in M_i$.

By construction, \bar{m} falls into Rule 1, and so $g(\bar{m}) = f(\theta)$. Fix any $i \in \mathcal{I}$ and any

 $m_i \in M_i$. Note that no deviation of i can induce Rule 4. Thus, Rules 1-3 apply.

- (A) Suppose that (m_i, \bar{m}_{-i}) falls into Rule 2. Then, $g(m) = e(m_i^1, \bar{m}_{-i}^1)$, and so $e(m_i^1, \theta) \in SL_i(f(\theta), \theta)$, by (22) of Lemma 8.
- (B) Suppose that (m_i, \bar{m}_{-i}) falls into Rule 3. Suppose that $u_i(f(\theta), \theta) \ge u_i(m_i^3(\theta), \theta)$. Then, $g(m_i, \bar{m}_{-i}) = \left(\frac{m_i^2}{m_i^2+1}\right) m_i^3(\theta) + \left(\frac{1}{m_i^2+1}\right) z_i(\theta, \theta)$. Since $\theta \in \bar{\Theta}_i^f$, the inequality in (20) of Lemma 7 implies that $z_i(\theta, \theta) \in SL_i(f(\theta), \theta)$. Since $u_i(f(\theta), \theta) \ge u_i(m_i^3(\theta), \theta)$, it follows that $g(m_i, \bar{m}_{-i}) \in SL_i(f(\theta), \theta)$. Suppose that $u_i(m_i^3(\theta), \theta) > u_i(f(\theta), \theta)$. Then, $g(m_i, \bar{m}_{-i}) = z_i(\theta, \theta) \in SL_i(f(\theta), \theta)$ given that $\theta \in \bar{\Theta}_i^f$. Therefore, in either case, $g(m_i, \bar{m}_{-i}) \in SL_i(f(\theta), \theta)$
- (C) Suppose that (m_i, \bar{m}_{-i}) falls into Rule 1. Then, there exists $\bar{\theta}' \in \Theta$ such that $\bar{m}_{-i}^1 \in \beta_{-i}^* (\bar{\theta}'), m_i^1 \in \beta_i^* (\bar{\theta}'), m_i^2 = 0$ if $i \in \mathcal{I}^{\beta^*(\bar{\theta}')}$, and $g(m_i, \bar{m}_{-i}) = f(\bar{\theta}')$. Lemma 11 implies that $u_i(f(\theta), \theta) \ge u_i(f(\bar{\theta}'), \theta)$.

Since $i \in \mathcal{I}$ and $m_i \in M_i$ are arbitrary, we conclude that the inequality in (34) is satisfied for all $i \in \mathcal{I}$ and all $m_i \in M_i$. Thus, $\overline{m} \in NE(\mathcal{M}, \theta)$. Since \overline{m} is also a rationalizable strategy profile at θ , it follows that $S^{\mathcal{M},\theta}$ is a nonempty set. According to Definition 1, to complete the proof, we need to show that $m \in S^{\mathcal{M},\theta}$ $\implies g(m) = f(\theta)$. To this end, we need additional notation.

For all $i \in \mathcal{I}$, let M_{-i}^0 be defined by

$$M_{-i}^{0} = \left\{ m_{-i} \in M_{-i} | m_{-i}^{1} \in \bigcup_{\bar{\theta} \in \Theta} \left(\bigcup_{\substack{\beta^{*}_{-i}(\bar{\theta}) \cap \Theta_{i}^{f} \neq \emptyset}} \beta^{*}_{-i}(\bar{\theta}) \right) \text{ and } m_{-i}^{2} = 0 \right\}.$$
(35)

Observe that M_{-i}^0 may be empty. Let $M_{-i}^0 \neq \emptyset$ and let us take any $m_{-i} \in M_{-i}^0$. Then, there exists $\bar{\theta} \in \Theta$ such that $m_{-i}^1 \in \beta_{-i}^*(\bar{\theta})$ and $\beta_{-i}^*(\bar{\theta}) \cap \Theta_i^f \neq \emptyset$. Since $\beta_{-i}^*(\bar{\theta}) \cap \Theta_i^f \neq \emptyset$, it follows from (31) that $i \notin \mathcal{I}^{\beta^*(\bar{\theta})}$. Part (iv) of Lemma 1 implies that $\beta_i^*(\bar{\theta}) = \Theta$. Thus, any strategy $(m_i, m_{-i}) \in M_i \times M_{-i}^0$ falls into Rule 1. For each $i \in \mathcal{I}$, let $\hat{m}_i = (\theta, 0, x_i(\cdot, \theta), y_i^*(\theta)) \in M_i$, where $x_i(\cdot, \theta)$ is defined in (30) and $y_i^*(\theta)$ is defined in (19). The following lemmata will help us to complete the proof.

Lemma 12. For all $i \in \mathcal{I}$, all $\theta \in \overline{\Theta}_i^f$, all $m_i \in M_i$ and all $\lambda_i^{\theta} \in \Delta (M_{-i} \setminus M_{-i}^0)$, if $i \in \mathcal{I}^{\beta^*(\theta)}$ and m_i is a best-response to λ_i^{θ} at θ , then $m_i^2 = 0.2^{22}$

Proof. Fix any $i \in \mathcal{I}$, any $\theta \in \bar{\Theta}_i^f$, and any $\lambda_i^{\theta} \in \Delta \left(M_{-i} \setminus M_{-i}^0 \right)$ and any $m_i \in M_i$ so that m_i is a best-response to λ_i^{θ} at θ . Recall that $\mathcal{I}^{\Theta} \neq \emptyset$. Assume, to the contrary, $m_i^2 > 0$. Since $m_i^2 > 0$ and $i \in \mathcal{I}^{\beta^*(\theta)}$, it follows that for any $m_{-i} \in supp(\lambda_i^{\theta}), (m_i, m_{-i})$ falls either into Rule 3 or into Rule 4. A contradiction of the initial assumption that m_i is a best response to λ_i^{θ} is derived if we show that $\hat{m}_i = (\theta, \hat{m}_i^2, x_i(\cdot, \theta), y_i^*(\theta))$ strictly dominates m_i .²³

Case 1: (m_i, m_{-i}) falls into Rule 3

Then, for some $\bar{\theta} \in \bar{\Theta}_i^f$, $(m_j^1, m_j^2) = (\bar{\theta}, 0)$ for all $j \in \mathcal{I} \setminus \{i\}$. Observe that if $u_i (f(\bar{\theta}), \bar{\theta}) \ge u_i (m_i^3(\bar{\theta}), \bar{\theta})$, then $u_i (x_i (\bar{\theta}, \theta), \theta) \ge u_i (m_i^3(\bar{\theta}), \theta)$ because $x_i (\bar{\theta}, \theta) \in B_i (\bar{\theta}, \theta)$, by (30). Since $\bar{\theta}, \theta \in \bar{\Theta}_i^f$, Lemma 9 implies that $u_i (x_i (\bar{\theta}, \theta), \theta) > u_i (z_i (\bar{\theta}, \bar{\theta}), \theta)$. By choosing an appropriate integer $\hat{m}_i^2 > 0$ and by changing m_i into \hat{m}_i , player *i* can induce Rule 3 with $u_i (f(\bar{\theta}), \bar{\theta}) \ge u_i (x_i (\bar{\theta}, \theta), \bar{\theta})$, obtain $g(\hat{m}_i, m_{-i})$ and be strictly better off at θ since $u_i (x_i (\bar{\theta}, \theta), \theta) > u_i (z_i (\bar{\theta}, \bar{\theta}), \theta)$.

Case 2: (m_i, m_{-i}) falls into Rule 4

Suppose that $j \in \mathcal{I}$ is the pivotal player. By definition of $y_i^*(\theta)$ in (19), $u_i(y_i^*(\theta), \theta) \ge u_i(m_j^4, \theta)$. Since $\theta \in \bar{\Theta}_i^f$, Lemma 6 implies that $u_i(y_i^*(\theta), \theta) > u_i(\underline{y}, \theta)$. By choosing an appropriate integer $\hat{m}_i^2 > 0$ and by changing m_i into \hat{m}_i , player *i* can induce Rule

²²This lemma holds only in the case $\mathcal{I}^{\Theta} \neq \emptyset$.

²³ \hat{m}_i strictly dominates m_i if $u_i(g(\hat{m}_i, m_{-i}), \theta) > u_i(g(m_i, m_{-i}), \theta)$ for all $m_{-i} \in M_{-i}$.

4, obtain $g(\hat{m}_i, m_{-i})$ and be strictly better off at θ since $u_i(y_i^*(\theta), \theta) > u_i(\underline{y}, \theta)$.

Since Θ is finite and since the choice of $m_{-i} \in supp(\lambda_i^{\theta})$ is arbitrary, we see that \hat{m}_i strictly dominates m_i by an appropriate choice of $\hat{m}_i^2 > 0$.

Lemma 13. For all $i \in \mathcal{I}$, all $\theta \in \overline{\Theta}_i^f$, all $m_i \in M_i$ and all $\lambda_i^{\theta} \in \Delta(M_{-i})$, if $i \in \mathcal{I}^{\beta^*(\theta)}$ and m_i is a best-response to λ_i^{θ} at θ , then there exists $m_{-i} \in supp(\lambda_i^{\theta})$ such that m_i is a best-response to m_{-i} at θ .²⁴

Proof. Fix any $i \in \mathcal{I}$, any $\theta \in \overline{\Theta}_i^f$, any $\lambda_i^{\theta} \in \Delta(M_{-i})$ and any $m_i \in M_i$ such that m_i is a best-response to λ_i^{θ} at θ . Assume, to the contrary, that m_i is not a best-response to any $m_{-i} \in supp(\lambda_i^{\theta})$ at θ . We proceed according to whether $supp(\lambda_i^{\theta}) \cap M_{-i}^0 \neq \emptyset$ or not.

Suppose that $supp(\lambda_i^{\theta}) \cap M_{-i}^0 \neq \emptyset$. Take any $m_{-i} \in supp(\lambda_i^{\theta}) \cap M_{-i}^0$. By (35), $m_{-i}^2 = 0$ and $m_{-i}^1 \in \bigcap_{j \in \mathcal{I} \setminus \{i\}} \beta_j^*(\bar{\theta})$ for some $\bar{\theta} \in \Theta$ such that $\left(\bigcap_{j \in \mathcal{I} \setminus \{i\}} \beta_j^*(\bar{\theta})\right) \cap \Theta_i^f \neq \emptyset$. Since $\left(\bigcap_{j \in \mathcal{I} \setminus \{i\}} \beta_j^*(\bar{\theta})\right) \cap \Theta_i^f \neq \emptyset$, it follows from (31) that $i \notin \mathcal{I}^{\beta^*(\bar{\theta})}$. Lemma ?? implies that $\beta_i^*(\bar{\theta}) = \Theta$. Thus, (m_i, m_{-i}) falls into Rule 1 and $g(m_i, m_{-i}) = f(\bar{\theta})$. Since i can induce only Rule 1 by changing his strategy, it follows that m_i is a bestresponse to $m_{-i} \in supp(\lambda_i^{\theta}) \cap M_{-i}^0$ at θ , which is a contradiction.

Suppose that $supp(\lambda_i^{\theta}) \cap M_{-i}^0 = \emptyset$. Since $\lambda_i^{\theta} \in \Delta(M_{-i} \setminus M_{-i}^0)$, Lemma 12 implies that $m_i^2 = 0$. A contradiction of the initial assumption that m_i is a best response to λ_i^{θ} is derived if we show that $\hat{m}_i = (\theta, \hat{m}_i^2, x_i(\cdot, \theta), y_i^*(\theta))$ strictly dominates m_i . To this end, fix any $m_{-i} \in supp(\lambda_i^{\theta})$. Since $m_{-i} \notin M_{-i}^0$, it follows from the definition of M_{-i}^0 in (35) that $m_j^2 \neq 0$ for some $j \in \mathcal{I} \setminus \{i\}$ or $m_{-i}^1 \notin \beta_{-i}^*(\tilde{\theta})$ for all $\tilde{\theta} \in \Theta$ such that $\left(\bigcap_{j \in \mathcal{I} \setminus \{i\}} \beta_j^*(\tilde{\theta})\right) \cap \Theta_i^f \neq \emptyset$. We proceed according to whether (m_i, m_{-i}) falls either

²⁴ m_i is a best-response to m_{-i} at θ if $u_i(g(m_i, m_{-i}), \theta) \ge u_i(g(m'_i, m_{-i}), \theta)$ for all $m'_i \in M_i$.

into Rule 1, or into Rule 2, or into Rule 3, or into Rule 4.

Case 1: (m_i, m_{-i}) falls into Rule 1

Then, there exists $\bar{\theta}$ such that $(m_j^1)_{j\in\mathcal{I}} \in \beta^*(\bar{\theta})$ and $m_j^2 = 0$ for all $j \in \mathcal{I}^{\beta^*(\bar{\theta})}$ and $g(m_i, m_{-i}) = f(\bar{\theta})$. Since $m_{-i} \notin M^0_{-i}$ and (m_i, m_{-i}) falls into Rule 1, it cannot be that $i \notin \mathcal{I}^{\beta^*(\bar{\theta})}$. Thus, let $i \in \mathcal{I}^{\beta^*(\bar{\theta})}$. We proceed according to whether $m_j^2 \neq 0$ for some $j \in \mathcal{I} \setminus \{i\}$ or not.

Sub-case 1.1: $m_j^2 = 0$ for all $j \in \mathcal{I} \setminus \{i\}$

Let us proceed according to whether there exists $\theta' \in \overline{\Theta}_i^f$ such that $m_j^1 = \theta'$ for all $j \in \mathcal{I} \setminus \{i\}$ or not.

Sub-case 1.1.1: For some $\theta' \in \bar{\Theta}_i^f$, $m_j^1 = \theta'$ for all $j \in \mathcal{I} \setminus \{i\}$

Since m_i is not a best-response to m_{-i} at θ , it holds that $L_i(f(\theta'), \theta') \cap SU_i(f(\bar{\theta}), \theta) \neq \emptyset$. \emptyset . To see it, assume, to the contrary, that $L_i(f(\theta'), \theta') \cap SU_i(f(\bar{\theta}), \theta) = \emptyset$, and so $L_i(f(\theta'), \theta') \subseteq L_i(f(\bar{\theta}), \theta)$. Note that player *i* cannot induce Rule 4 since $m_j^1 = \theta' \in \bar{\Theta}_i^f$ for all $j \in \mathcal{I} \setminus \{i\}$. Take any $\tilde{m}_i \in M_i$.

- If (\tilde{m}_i, m_{-i}) falls into Rule 2, then $g(\tilde{m}_i, m_{-i}) = e(\tilde{m}_i^1, \theta') \in SL_i(f(\theta'), \theta')$.
- If (\tilde{m}_i, m_{-i}) falls into Rule 3, then $g(\tilde{m}_i, m_{-i}) = \left(\frac{\tilde{m}_i^2}{\tilde{m}_i^2 + 1}\right) \tilde{m}_i^3(\theta') + \left(\frac{1}{\tilde{m}_i^2 + 1}\right) z_i(\theta', \theta')$ if $u_i(f(\theta'), \theta') \ge u_i(m_i^3(\theta'), \theta')$, otherwise, $g(\tilde{m}_i, m_{-i}) = z_i(\theta', \theta')$. Since $z_i(\theta', \theta') \in L_i(f(\theta'), \theta')$ by (20) and since $u_i(f(\theta'), \theta') \ge u_i(m_i^3(\theta'), \theta')$, it follows that $g(\tilde{m}_i, m_{-i}) \in L_i(f(\theta'), \theta')$.
- If (\tilde{m}_i, m_{-i}) falls into Rule 1, then there exists $\bar{\theta}'$ such that $(\tilde{m}_i^1, m_{-i}^1) \in \beta^*(\bar{\theta}')$, $m_j^2 = 0$ for all $j \in \mathcal{I}^{\beta^*(\bar{\theta}')}$ and $g(\tilde{m}_i, m_{-i}) = f(\bar{\theta}')$. By arguing as above, we can see that $(m'_i, m_{-i}) \in NE(\Gamma, \theta')$, where $m'_i = (\theta', 0, m_i^3, m_i^4)$ and $m_{-i} = (\theta', 0, m_{-i}^3, m_{-i}^4)$, that (m'_i, m_{-i}) falls into Rule 1 and that $g(m'_i, m_{-i}) = f(\theta')$. This implies that $g(\tilde{m}_i, m_{-i}) = f(\bar{\theta}') \in L_i(f(\theta'), \theta')$.

Since $L_i(f(\theta'), \theta') \subseteq L_i(f(\bar{\theta}), \theta)$, it follows that $g(\tilde{m}_i, m_{-i}) \in L_i(f(\bar{\theta}), \theta)$. Since the choice of $\tilde{m}_i \in M_i$ is arbitrary, we have that m_i is a best-response to m_{-i} , which is a contradiction. Thus, $L_i(f(\theta'), \theta') \cap SU_i(f(\bar{\theta}), \theta) \neq \emptyset$.

Since $L_i(f(\theta'), \theta') \cap SU_i(f(\bar{\theta}), \theta) \neq \emptyset$, it follows that $\alpha_i^{\theta}(\theta', \bar{\theta}) \in L_i(f(\theta'), \theta') \cap$ $SU_i(f(\bar{\theta}), \theta)$ by definition of $\alpha_i^{\theta}(\theta', \bar{\theta})$ given in (28). Since $\theta \in \bar{\Theta}_i^f$, requirement (iii) of the definition of $\bar{D}_i(\theta', \theta)$ implies that $\alpha_i^{\theta}(\theta', \bar{\theta}) \in \bar{D}_i(\theta', \theta)$.

We show that $u_i(x_i(\theta',\theta),\theta) > u_i(f(\bar{\theta}),\theta)$. Let us proceed according to whether $D_i(\theta',\theta) \neq \emptyset$ or not.

- Suppose that $D_i(\theta',\theta) \neq \emptyset$. Then, $L_i(f(\theta'),\theta') \cap SU_i(f(\theta'),\theta) \neq \emptyset$ by definition of $D_i(\theta',\theta)$ in (27). Since $\emptyset \neq \overline{D}_i(\theta',\theta) \subseteq D_i(\theta',\theta)$ by definition and, moreover, since $x_i(\theta',\theta) \in \arg \max_{y \in \overline{D}_i(\theta',\theta)} u_i(y,\theta)$ by (29)-(30), it follows that $u_i(x_i(\theta',\theta),\theta) > u_i(f(\theta'),\theta)$. Clearly, $u_i(x_i(\theta',\theta),\theta) > u_i(f(\overline{\theta}),\theta)$ if $u_i(f(\theta'),\theta) \geq u_i(f(\overline{\theta}),\theta)$. Otherwise, let $u_i(f(\overline{\theta}),\theta) > u_i(f(\theta'),\theta)$. Since $\alpha_i^{\theta}(\theta',\overline{\theta}) \in \overline{D}_i(\theta',\theta) \cap SU_i(f(\overline{\theta}),\theta)$ and since $u_i(x_i(\theta',\theta),\theta) \geq u_i(\alpha_i^{\theta}(\theta',\overline{\theta}),\theta)$, it follows that $u_i(x_i(\theta',\theta),\theta) > u_i(f(\overline{\theta}),\theta) > u_i(f(\overline{\theta}),\theta)$.
- Suppose that $D_i(\theta',\theta) = \emptyset$. Then, $L_i(f(\theta'),\theta') \subseteq L_i(f(\theta'),\theta)$. By definition of $B_i(\theta',\theta)$ in (29) and by definition of $x_i(\cdot,\theta)$ in (30), $x_i(\theta',\theta) = f(\theta')$. Assume, to the contrary, that $u_i(f(\bar{\theta}),\theta) \ge u_i(f(\theta'),\theta)$. We have already established above that $\alpha_i^{\theta}(\theta',\bar{\theta}) \in L_i(f(\theta'),\theta') \cap SU_i(f(\bar{\theta}),\theta)$. Thus, $u_i(f(\theta'),\theta') \ge u_i(\alpha_i^{\theta}(\theta',\bar{\theta}),\theta')$ and $u_i(\alpha_i^{\theta}(\theta',\bar{\theta}),\theta) > u_i(f(\theta'),\theta)$. This implies that $D_i(\theta',\theta) \neq \emptyset$, which is a contradiction. Thus, $u_i(x_i(\theta',\theta),\theta) > u_i(f(\bar{\theta}),\theta)$ where $x_i(\theta',\theta) = f(\theta')$.

We conclude that $u_i(x_i(\theta',\theta),\theta) > u_i(f(\bar{\theta}),\theta)$. Since $m_j^2 = 0$ and $m_j^1 = \theta' \in \bar{\Theta}_i^f$ for all $j \in \mathcal{I} \setminus \{i\}$ and $i \in \mathcal{I}^{\beta^*(\bar{\theta})}$, by choosing an appropriate integer $\hat{m}_i^2 > 0$ and by changing m_i into \hat{m}_i , player *i* can induce Rule 3 with $u_i(f(\theta'), \theta') \ge u_i(x_i(\theta', \theta), \theta')$, obtain $g(\hat{m}_i, m_{-i})$ and be strictly better off at θ since $u_i(x_i(\theta', \theta), \theta) > u_i(f(\bar{\theta}), \theta)$.

Sub-case 1.1.2: There does not exist any $\theta' \in \bar{\Theta}_i^f$ such that $m_j^1 = \theta'$ for all $j \in \mathcal{I} \setminus \{i\}$ Since $m_j^2 = 0$ for all $j \in \mathcal{I} \setminus \{i\}$ and $m_{-i} \notin M_{-i}^0$, it follows from the definition of M_{-i}^0 in (35) that $m_{-i}^1 \notin \beta_{-i}^* \left(\tilde{\theta}\right)$ for all $\tilde{\theta} \in \Theta$ such that $\left(\bigcap_{j \in \mathcal{I} \setminus \{i\}} \beta_j^* \left(\tilde{\theta}\right)\right) \cap \Theta_i^f \neq \emptyset$. Suppose that I = 2. Then, $m_{-i}^1 \notin \bar{\Theta}_i^f$. Since $\theta' \in \beta_{-i}^* (\theta')$ for all $\theta' \in \Theta_i^f$ and since $m_{-i}^1 \notin \beta_{-i}^* \left(\tilde{\theta}\right)$ for all $\tilde{\theta} \in \Theta$ such that $\beta_{-i}^* \left(\tilde{\theta}\right) \cap \Theta_i^f \neq \emptyset$, it follows that $m_{-i}^1 \in \bar{\Theta}_i^f$, which is a contradiction. Thus, let us suppose that $I \geq 3$.

Suppose that there exists $j, j' \in \mathcal{I} \setminus \{i\}$, with $j \neq j'$, such that $m_j^1 \neq m_{j'}^1$. By definition of $y_i^*(\theta)$ in (19), $u_i(y_i^*(\theta), \theta) \geq u_i(f(\bar{\theta}), \theta)$. Since $\theta \in \bar{\Theta}_i^f$, Lemma 6 implies that $u_i(y_i^*(\theta), \theta) > u_i(\underline{y}, \theta)$. Suppose that $u_i(y_i^*(\theta), \theta) = u_i(f(\bar{\theta}), \theta)$. Then, by definition of g, we have that m_i is a best-response to m_{-i} at θ , which is a contradiction. Suppose that $u_i(y_i^*(\theta), \theta) > u_i(f(\bar{\theta}), \theta)$. By choosing an appropriate integer $\hat{m}_i^2 > 0$ and by changing m_i into \hat{m}_i , player i can induce Rule 4, obtain $g(\hat{m}_i, m_{-i})$ and be strictly better off at θ since $u_i(y_i^*(\theta), \theta) > u_i(f(\bar{\theta}), \theta)$.

Suppose that for all $j, j' \in \mathcal{I} \setminus \{i\}, m_j^1 = m_{j'}^1$. Since there does not exist any $\theta' \in \bar{\Theta}_i^f$ such that $m_j^1 = \theta'$ for all $j \in \mathcal{I} \setminus \{i\}$, it follows that $m_j^1 = \tilde{\theta} \in \Theta_i^f$ and $m_j^1 = m_{j'}^1 = \tilde{\theta}$ for all $j, j' \in \mathcal{I} \setminus \{i\}$. Since $m_{-i}^1 \in \beta_{-i}^*(\bar{\theta})$, we have that $m_j^1 = \tilde{\theta} \in \bigcap_{j \in \mathcal{I} \setminus \{i\}} \beta_j^*(\bar{\theta})$. However, since $i \in \mathcal{I}^{\beta^*(\bar{\theta})}$, it follows from (31) that $m_j^1 = \tilde{\theta} \notin \Theta_i^f$, which is a contradiction.

Sub-case 1.2: $m_j^2 \neq 0$ for some $j \in \mathcal{I} \setminus \{i\}$

Since $i \in \mathcal{I}^{\beta^*(\bar{\theta})}$, by choosing an appropriate integer $\hat{m}_i^2 > 0$ and by changing m_i into \hat{m}_i , player *i* can induce Rule 4. Suppose that $u_i(y_i^*(\theta), \theta) = u_i(f(\bar{\theta}), \theta)$. Then, by definition of *g*, we have that m_i is a best-response to m_{-i} at θ , which is a contradiction. Suppose that $u_i(y_i^*(\theta), \theta) > u_i(f(\bar{\theta}), \theta)$. Player *i* can obtain $g(\hat{m}_i, m_{-i})$ and be strictly better off at θ since $u_i(y_i^*(\theta), \theta) > u_i(f(\bar{\theta}), \theta)$.

Case 2: (m_i, m_{-i}) falls into Rule 2

Then, $m_1^2 = m_2^2 = 0$, $(m_i^1)_{i \in \mathcal{I}} \notin \beta^* \left(\overline{\theta}\right)$ for all $\overline{\theta} \in \Theta$ and $g(m) = e\left(m_i^1, m_{-i}^1\right)$. Recall that $\theta \in \overline{\Theta}_i^f$. Then, to apply Lemma 10, we need to guarantee that $m_{-i}^1 \in \overline{\Theta}_i^f$. Since $m_{-i} \notin M_{-i}^0$ and $m_{-i}^2 = 0$, $m_{-i}^1 \notin \beta_{-i}^* \left(\overline{\theta}\right)$ for all $\overline{\theta} \in \Theta$ such that $\beta_{-i}^* \left(\overline{\theta}\right) \cap \Theta_i^f \neq \emptyset$, by (35). Since $\theta' \in \beta_{-i}^* (\theta')$ for all $\theta' \in \Theta_i^f$, it follows that $m_{-i}^1 \notin \Theta_i^f$, and so $m_{-i}^1 \in \overline{\Theta}_i^f$. Since $\theta, m_{-i}^1 \in \overline{\Theta}_i^f$, Lemma 10 implies that $u_i \left(x_i \left(m_{-i}^1, \theta\right), \theta\right) > u_i \left(e \left(m_i^1, m_{-i}^1\right), \theta\right)$.

Suppose that *i* changes m_i into \hat{m}_i in which $\hat{m}_i^2 > 0$. Then, (\hat{m}_i, m_{-i}) cannot fall into Rule 1. To see it, assume, on the contrary, that it falls into Rule 1. Then, there exists $\bar{\theta}'$ such that $(\hat{m}_i^1, m_{-i}^1) \in \beta_i^*(\bar{\theta}') \times \beta_{-i}^*(\bar{\theta}')$. Since $\hat{m}_i^2 > 0$ and (\hat{m}_i, m_{-i}) falls into Rule 1, it follows that $\mathcal{I}^{\beta^*(\bar{\theta}')} = \{-i\}$ given that $\mathcal{I}^{\Theta} \neq \emptyset$. Since $i \notin \mathcal{I}^{\beta^*(\bar{\theta}')}$, it follows from (31) that $\beta_{-i}^*(\bar{\theta}') \cap \Theta_i^f \neq \emptyset$. Since $m_{-i} \notin M_{-i}^0$ and $m_{-i}^2 = 0$ and since $\beta_{-i}^*(\bar{\theta}') \cap \Theta_i^f \neq \emptyset$, (35) implies that $m_{-i}^1 \notin \beta_{-i}^*(\bar{\theta}')$, which is a contradiction. Thus, (\hat{m}_i, m_{-i}) cannot fall into Rule 1.

Since $m_{-i}^1 \in \bar{\Theta}_i^f$, by choosing an appropriate integer $\hat{m}_i^2 > 0$ and by changing m_i into \hat{m}_i , player *i* can induce Rule 3 with $u_i \left(f\left(m_{-i}^1 \right), m_{-i}^1 \right) \ge u_i \left(x_i \left(m_{-i}^1, \theta \right), m_{-i}^1 \right)$ -by definition of $x_i \left(m_{-i}^1, \theta \right)$ in (30), obtain $g \left(\hat{m}_i, m_{-i} \right)$ and be strictly better off at θ since $u_i \left(x_i \left(m_{-i}^1, \theta \right), \theta \right) > u_i \left(e \left(m_i^1, m_{-i}^1 \right), \theta \right)$.

Case 3: (m_i, m_{-i}) falls into Rule 3

Then, for some $h \in \mathcal{I}$ and some $\bar{\theta} \in \bar{\Theta}_h^f$, $(m_j^1, m_j^2) = (\bar{\theta}, 0)$ for all $j \in \mathcal{I} \setminus \{h\}$, and either $m_h^2 > 0$ or $[m_h^2 = 0, I \neq 2$ and $(m_k^1)_{k \in \mathcal{I}} \notin \beta^* (\hat{\theta})$ for all $\hat{\theta} \in \Theta$]. Let us proceed according to whether $m_i^1 = \bar{\theta}$ or not.

Sub-case 3.1: $m_i^1 = \bar{\theta}$.

Since $m_i^2 = 0$, it follows that $h \neq i$. Since $\theta \in \overline{\Theta}_i^f$, Lemma 6 implies that $u_i(y_i^*(\theta), \theta) > u_i(\underline{y}, \theta)$. Moreover, by definition of $y_i^*(\theta)$ in (19), it holds that

 $u_i(y_i^*(\theta), \theta) \ge u_i(m_h^3(\bar{\theta}), \theta)$ and $u_i(y_i^*(\theta), \theta) \ge u_i(z_h(\bar{\theta}, \bar{\theta}), \theta)$. We proceed according to whether $u_i(y_i^*(\theta), \theta) = u_i(m_h^3(\bar{\theta}), \theta)$ and $u_i(y_i^*(\theta), \theta) = u_i(z_h(\bar{\theta}, \bar{\theta}), \theta)$ or not.

- Suppose that $u_i(y_i^*(\theta), \theta) = u_i(m_h^3(\bar{\theta}), \theta)$ and $u_i(y_i^*(\theta), \theta) = u_i(z_h(\bar{\theta}, \bar{\theta}), \theta)$. Since *i* cannot be strictly better off by changing m_i , it follows that m_i is a best-response to m_{-i} , which is a contradiction.
- Suppose that u_i (y_i^{*}(θ), θ) > u_i (m_h³(θ), θ) or u_i (y_i^{*}(θ), θ) > u_i (z_h (θ, θ), θ). By choosing an appropriate integer m̂_i² > 0 and by changing m_i into m̂_i, player i can induce Rule 4, obtain g(m̂_i, m_{-i}) and be strictly better off at θ since u_i (y_i^{*}(θ), θ) > u_i (m_h³(θ), θ) or u_i (y_i^{*}(θ), θ) > u_i (z_h (θ, θ), θ).

Sub-case 3.2: $m_i^1 \neq \bar{\theta}$.

Since $m_i^2 = 0$, we have that either $i \neq h$ if I = 2 or i = h if $I \neq 2$. If I = 2, then ican find a profitable deviation by changing m_i into \hat{m}_i -by arguing as in sub-case 3.1. Suppose that $I \neq 2$. Then, $g(m) = z_i(\bar{\theta}, \bar{\theta})$ because $m_i^2 = 0$. Since $\bar{\theta}, \theta \in \bar{\Theta}_i^f$, Lemma 9 implies that $u_i(x_i(\bar{\theta}, \theta), \theta) > u_i(z_i(\bar{\theta}, \bar{\theta}), \theta)$. By choosing an appropriate integer $\hat{m}_i^2 > 0$ and by changing m_i into \hat{m}_i , player i can induce Rule 3, obtain $g(\hat{m}_i, m_{-i})$ and be strictly better off at θ since $u_i(x_i(\bar{\theta}, \theta), \theta) > u_i(z_i(\bar{\theta}, \bar{\theta}), \theta)$.

Case 4: (m_i, m_{-i}) falls into Rule 4

Suppose that j is the pivotal player. By definition of $y_i^*(\theta)$ in (19), $u_i(y_i^*(\theta), \theta) \ge u_i(m_j^4, \theta)$. Recall that $m_i^2 = 0$. We proceed according to whether $m_h^2 \neq 0$ for some $h \in \mathcal{I} \setminus \{i\}$ or not.

Sub-case 4.1: $m_h^2 \neq 0$ for some $h \in \mathcal{I} \setminus \{i\}$

Since $\theta \in \bar{\Theta}_i^f$, Lemma 6 implies that $u_i(y_i^*(\theta), \theta) > u_i(\underline{y}, \theta)$. By choosing an appropriate integer $\hat{m}_i^2 > 0$ and by changing m_i into \hat{m}_i , player *i* can induce Rule 4,

obtain $g(\hat{m}_i, m_{-i})$ and be strictly better off at θ since $u_i(y_i^*(\theta), \theta) > u_i(y, \theta)$.

Sub-case 4.2: $m_h^2 = 0$ for all $h \in \mathcal{I} \setminus \{i\}$

Thus, $m_i^2 = m_h^2 = 0$ for all $h \in \mathcal{I} \setminus \{i\}$ and $g(m) = \underline{y}$. Since (m_i, m_{-i}) does not fall into Rule 1 and $m_k^2 = 0$ for all $k \in \mathcal{I}$, it holds that $(m_k^1)_{k \in \mathcal{I}} \notin \beta^*(\overline{\theta})$ for all $\overline{\theta} \in \Theta$. If I = 2, then (m_i, m_{-i}) falls into Rule 2, which is a contradiction. Therefore, let $I \neq 2$. Since (m_i, m_{-i}) falls into Rule 4 and $m_i^2 = m_h^2 = 0$ for all $h \in \mathcal{I} \setminus \{i\}$, it cannot be that there exists $\overline{\theta} \in \Theta$ such that $(m_j^1)_{j \in \mathcal{I}} \in \beta^*(\overline{\theta})$ -otherwise, m would fall into Rule 1. Therefore, $(m_j^1)_{j \in \mathcal{I}} \notin \bigcup_{\overline{\theta} \in \Theta} \beta^*(\overline{\theta})$. Since (m_i, m_{-i}) falls into Rule 4 and $m_i^2 = m_h^2 = 0$ for all $h \in \mathcal{I} \setminus \{i\}$, it follows that either $m_k^1 = m_j^1$ for all $k, j \in \mathcal{I} \setminus \{i\}$ and $m_k^1 \notin \overline{\Theta}_i^f$ or there exists $k \in \mathcal{I} \setminus \{i\}$ such that $m_k^1 \neq m_j^1$ for some $j \in \mathcal{I} \setminus \{i\}$. In either case, by changing m_i into \hat{m}_i with $\hat{m}_i^2 > 0$, player i induces Rule 4. Since $\theta \in \overline{\Theta}_i^f$, Lemma 6 implies that $u_i(y_i^*(\theta), \theta) > u_i(\underline{y}, \theta)$. Then, by choosing an appropriate integer $\hat{m}_i^2 > 0$, i can obtain $g(\hat{m}_i, m_{-i})$ and be strictly better off at θ since $u_i(y_i^*(\theta), \theta) > u_i(y, \theta)$.

Since Θ is finite and since the choice of $m_{-i} \in supp(\lambda_i^{\theta})$ is arbitrary, we see that \hat{m}_i strictly dominates every element of $supp(\lambda_i^{\theta})$ by an appropriate choice of $\hat{m}_i^2 > 0$. **Lemma 14.** For all $i \in \mathcal{I}$, all $\theta \in \bar{\Theta}_i^f$, all $m_i \in M_i$ and all $m_{-i} \in M_{-i}$, if $i \in \mathcal{I}^{\beta^*(\theta)}$ and m_i is a best-response to m_{-i} at θ , then (m_i, m_{-i}) falls into Rule 1.

Proof. Fix any $i \in \mathcal{I}$, any $\theta \in \bar{\Theta}_i^f$, any $m_{-i} \in M_{-i}$ and any $m_i \in M_i$ such that m_i is a best-response to m_{-i} at θ . Suppose that $i \in \mathcal{I}^{\beta^*(\theta)}$. Assume, to the contrary, that (m_i, m_{-i}) does not fall into Rule 1. This implies that $(m_i^1, m_{-i}^1) \notin \beta^*(\bar{\theta})$ for all $\bar{\theta} \in \Theta$ or [for some $\ell \in \mathcal{I}$ such that $\ell \in \mathcal{I}^{\beta^*(\bar{\theta})}$, it holds that $m_{\ell}^2 \neq 0$, where $\bar{\theta} \in \Theta$ is such that $(m_i^1, m_{-i}^1) \in \beta^*(\bar{\theta})$].

Let us show that $m_{-i} \notin M^0_{-i}$. Suppose that $m_{-i} \in M^0_{-i}$. By (35), $m^2_{-i} = 0$ and $m^1_{-i} \in \beta^*_{-i}\left(\tilde{\theta}\right)$ for some $\tilde{\theta} \in \Theta$ such that $\beta^*_{-i}\left(\tilde{\theta}\right) \cap \Theta^f_i \neq \emptyset$. This implies that $i \notin \mathcal{I}^{\beta^*(\tilde{\theta})}$. Moreover, Lemma 1 implies that $\beta^*_i\left(\tilde{\theta}\right) = \Theta$. Since $m^1_{-i} \in \beta^*_{-i}\left(\tilde{\theta}\right)$ and

 $m_{-i}^2 = 0$, it follows that (m_i, m_{-i}) falls into Rule 1, which is a contradiction. Thus, $m_{-i} \notin M_{-i}^0$.

Since $i \in \mathcal{I}^{\beta^*(\theta)}$, Lemma 12 implies that $m_i^2 = 0$. Thus, (m_i, m_{-i}) falls either into Rule 2, or into Rule 3, or into Rule 4. By arguing as in Cases 2-4 of the proof of Lemma 13, we see that m_i is not a best-response to m_{-i} at θ , which is a contradiction.

For all $i \in \mathcal{I}$, let

$$E_i = \left\{ m_i^1 \in M_i^1 \middle| m_i \in S_i^{\mathcal{M},\theta} \right\}.$$
(36)

Since $\overline{m} \in NE(\mathcal{M}, \theta)$, it follows that $E_i \neq \emptyset$ for all $i \in \mathcal{I}$. Specifically, it follows that $\theta \in E_i$ for all $i \in \mathcal{I}$. Let us show that $E_i \subseteq \beta_i^*(\theta)$ for all $i \in \mathcal{I}$. Fix any $i \in \mathcal{I}$. Suppose that $\theta \in \Theta_i^f$. Part (iv) of Lemma 1 implies that $\beta_i^*(\theta) = \Theta$, and so $E_i \subseteq \beta_i^*(\theta)$, as we sought. Otherwise, let us suppose that $\theta \in \overline{\Theta}_i^f$. Since $\beta^* \in \mathcal{E}(\mathcal{R})$, it suffices to show that $E_i \subseteq \mathcal{R}_i^{\beta^*}(\theta)$.

Fix any $\ell \in \mathcal{I}$ and any $m_{\ell}^1 \in E_{\ell}$. (36) implies that there exists $m_{\ell} \in S_{\ell}^{\mathcal{M},\theta}$. Suppose that $\beta_{-\ell}^*(\theta) \cap \Theta_{\ell}^f \neq \emptyset$. Part (iv) of Lemma 1 implies that $\beta_{\ell}^*(\theta) = \Theta$. We can apply (5) to player ℓ under the specification that $\tilde{\theta} = \hat{\theta} = \theta$. Otherwise, let us suppose that $\beta_{-\ell}^*(\theta) \subseteq \bar{\Theta}_{\ell}^f$, so that $\ell \in \mathcal{I}^{\beta^*(\theta)}$. Since $m_{\ell} \in S_{\ell}^{\mathcal{M},\theta}$, it follows that m_{ℓ} is a best-response to some λ_{ℓ}^{θ} at θ . Since $\beta_{-\ell}^*(\theta) \subseteq \bar{\Theta}_{\ell}^f$, it follows that $M_{\ell}^0 = \emptyset$, and so $\lambda_{\ell}^{\theta} \in \Delta \left(M_{-\ell} \setminus M_{-\ell}^0\right)$. Lemma 12 implies that $m_{\ell}^2 = 0$, Lemma 13 implies that m_{ℓ} is a best-response to some $m_{-\ell} \in supp(\lambda_{\ell}^{\theta})$ at θ , and Lemma 14 implies that $(m_{\ell}, m_{-\ell})$ falls into Rule 1. Then, there exists $\bar{\theta}$ such that $m^1 \in \beta^*(\bar{\theta})$ and $g(m_{\ell}, m_{-\ell}) = f(\bar{\theta})$. By definition of $S_{\ell}^{\mathcal{M},\theta}$, it also follows that $m_{-\ell} \in S_{-\ell}^{\mathcal{M},\theta}$, and so $m_{-\ell}^1 \in E_{-\ell}$. Since m_{ℓ} is a best-response to some $m_{-\ell}$, it holds that $L_{\ell}\left(f\left(m_{-\ell}^1\right), m_{-\ell}^1\right) \subseteq L_{\ell}\left(f(\bar{\theta}), \theta\right)$. Thus, we can apply (5) to player ℓ under the specification that $\tilde{\theta} = \bar{\theta}$ and $\hat{\theta} = m_{-\ell}^1$.

Since the choice of $\ell \in \mathcal{I}$ is arbitrary, we have that (5) applies to every player $\ell \in \mathcal{I}$. It follows that $E_i \subseteq \mathcal{R}_i^{\beta^*}(\theta)$. Since the choice of $i \in \mathcal{I}$ was arbitrary, we have that

$$E_i \subseteq \mathcal{R}_i^{\beta^*}(\theta) = \beta_i^*(\theta) \tag{37}$$

for all $i \in \mathcal{I}$.

To complete the proof, let us show that $g(m) = f(\theta)$ for all $m \in S^{\mathcal{M},\theta}$. Fix any $m \in S^{\mathcal{M},\theta}$. (36) implies that $(m_i^1, m_{-i}^1) \in E_i \times E_{-i}$. (36) implies that $(m_i^1, m_{-i}^1) \in \beta^*(\theta)$. If $\mathcal{I}^{\beta^*(\theta)} = \emptyset$, then m falls into Rule 1 and that $g(m) = f(\theta)$. Suppose that $\mathcal{I}^{\beta^*(\theta)} \neq \emptyset$. Fix any $\ell \in \mathcal{I}^{\beta^*(\theta)}$. Since $m_\ell \in S_\ell^{\mathcal{M},\theta}$, it follows that m_ℓ is a best-response to some λ_ℓ^{θ} at θ . Since $\beta_{-\ell}^*(\theta) \subseteq \overline{\Theta}_\ell^f$, it follows that $M_\ell^0 = \emptyset$, and so $\lambda_\ell^\theta \in \Delta \left(M_{-\ell} \setminus M_{-\ell}^0\right)$. Lemma 12 implies that $m_\ell^2 = 0$. Since the choice of $\ell \in \mathcal{I}^{\beta^*(\theta)}$ is arbitrary, it follows that $m_\ell^2 = 0$ for all $\ell \in \mathcal{I}^{\beta^*(\theta)}$. Therefore, m falls into Rule 1 and $g(m) = f(\theta)$. Since the choice of $m \in S^{\mathcal{M},\theta}$ was arbitrary, it follows that $g(m) = f(\theta)$ for all $m \in S^{\mathcal{M},\theta}$.

C. PROOF OF THEOREM 2

In this Appendix, we study the relationship between IM and SEM^{**}. The implementing condition of Xiong (2022) termed Strict Event Monotonicity^{**} with respect to the partition P. To introduce SEM^{**}, we need additional notation. For each $\theta \in \Theta$, let us define \mathcal{I}^{θ} by $\mathcal{I}^{\theta} = \{i \in \mathcal{I} | SL_i(f(\theta), \theta) \neq \emptyset\}$, and for each $E \in 2^{\Theta} \setminus \{\emptyset\}$, let us define \mathcal{I}^E by $\mathcal{I}^E = \bigcap_{\theta \in E} \mathcal{I}^{\theta}$.

Definition 5. $f: \Theta \to Y$ is Strict Event Monotonic^{**} (henceforth, SEM^{**}) if there exists $P \in \mathcal{P}_f$ such that the following conditions are satisfied.

1. For all $(\theta', E) \in \Theta \times 2^{\Theta} \setminus \{\emptyset\},\$

for all
$$(i, \theta) \in \mathcal{I}^{\cup_{\theta \in E} P(\theta)} \times E$$
, there exists $\hat{\theta}(i) \in P(\theta)$
such that $SL_i\left(f(\theta), \hat{\theta}(i)\right) \subseteq L_i(f(\theta), \theta') \qquad \Rightarrow P(\theta') = \bigcup_{\theta \in E} P(\theta)$

2. For all $i \in \mathcal{I}$ and all $\theta, \theta', \hat{\theta} \in \Theta$,

$$\begin{bmatrix} \{i\} = \mathcal{I}^{\theta} \\ \text{and } P(\theta) \neq P(\theta') \end{bmatrix} \implies L_i\left(f\left(\hat{\theta}\right), \hat{\theta}\right) \cap SU_i\left(f\left(\theta\right), \theta'\right) \neq \emptyset$$

Before presenting this equivalence result, let us state and prove the following claim.

Lemma 15. Suppose f is IM and for any $\theta, \theta', \beta^*(\theta') \subseteq \beta^*(\theta)$, then $\beta^*(\theta') \subseteq \beta^*(\theta)$

Proof. Suppose f is IM and for some $\theta, \theta', \beta^*(\theta') \subseteq \beta^*(\theta)$. $\theta \in E$ and let us show that $\beta^*(\theta') \subseteq \beta^*(\theta)$. Again, since $\beta^* \in \mathcal{E}(\mathcal{R})$, it is sufficient to show that $\beta^*_i(\theta') \subseteq \mathcal{R}_i^{\beta^*}(\theta)$ for all $i \in \mathcal{I}$. Fix any $i \in \mathcal{I}$. Since we have already shown that $\beta^*(\theta) \subseteq \beta^*(\theta')$, it follows that $\theta \in \beta^*_j(\theta')$ for all $j \in \mathcal{I}$. Moreover, IM implies that $f(\theta') = f(\theta)$. Let $E'_j = \beta^*_j(\theta')$ for all $j \in \mathcal{I}$.

Fix any $\ell \in \mathcal{I}$ and any $\bar{\theta} \in E'_{\ell}$. Then, $\theta' \in E'_{\ell}$, $\theta \in \bigcap_{j \in \mathcal{I} \setminus \{\ell\}} E'_{j}$, $\bar{\theta} \in \beta^{*}_{\ell}(\theta')$, $\theta \in \bigcap_{j \in \mathcal{I} \setminus \{\ell\}} \beta^{*}_{j}(\theta')$ and $L_{\ell}(f(\theta), \theta) \subseteq L_{\ell}(f(\theta'), \theta)$. Since this holds for any $\ell \in \mathcal{I}$ and any $\bar{\theta} \in E'_{\ell}$, it follows from (5) that $\beta^{*}_{i}(\theta') = E'_{i} \subseteq \mathcal{R}^{\beta^{*}}_{i}(\theta)$. Since the choice of i was arbitrary, we conclude that $\beta^{*}(\theta') \subseteq \mathcal{R}^{\beta^{*}}(\theta)$.

PROOF OF THEOREM 2 Assume that $I \geq 3$ and that $\mathcal{I}^{\Theta} \neq \emptyset$. Suppose that f satisfies IM. We show that f is SEM^{**}. Let $P : \Theta \to 2^{\Theta} \setminus \{\emptyset\}$ be defined by

$$P(\theta) = \{\theta' \in \Theta | \beta^*(\theta) = \beta^*(\theta')\}.$$
(38)

Let us show that $P \in \mathcal{P}_f$. Since $\theta \in \beta^*(\theta) \subseteq P(\theta)$, it follows that $P(\theta) \neq \emptyset$ and $\cup_{\theta \in \Theta} P(\theta) = \Theta$. Fix any $\theta, \theta' \in \Theta$ and suppose that $P(\theta) \neq P(\theta')$. We show that $P(\theta) \cap P(\theta') = \emptyset$. Assume, to the contrary, that there exists $\overline{\theta} \in P(\theta) \cap P(\theta')$. By (38), $\beta^*(\theta) = \beta^*(\overline{\theta})$ and $\beta^*(\theta') = \beta^*(\overline{\theta})$, and so $\beta^*(\theta) = \beta^*(\theta')$. Therefore, $P(\theta) = P(\theta')$, which is a contraction. Thus, P is a partition of Θ . Finally, since fsatisfies IM, it can be seen that P is at least as fine as P_f . Thus, $P \in \mathcal{P}_f$.

Let us show that f satisfies part (1) of SEM^{**}. To this end, fix any $(\theta', E) \in \Theta \times 2^{\Theta} \setminus \{\emptyset\}$. Suppose that for all $(i, \theta) \in \mathcal{I}^{\cup_{\theta \in E} P(\theta)} \times E$, there exists $\hat{\theta}(i) \in P(\theta)$ such that $SL_i\left(f(\theta), \hat{\theta}(i)\right) \subseteq L_i(f(\theta), \theta')$. This is equivalent to the following statement: for all $(i, \theta) \in \mathcal{I}^{\cup_{\theta \in E} P(\theta)} \times E$, there exists $\hat{\theta}(i) \in P(\theta)$ such that $L_i\left(f(\theta), \hat{\theta}(i)\right) \subseteq L_i(f(\theta), \theta')$. We show that $P(\theta') = \bigcup_{\theta \in E} P(\theta)$. By (38), it is sufficient to show that

 $\beta^*(\theta) = \beta^*(\theta')$ for all $\theta \in E$. To this end, we first show that $\beta_i^*(E) \subseteq \beta_i^*(\theta')$ for all $i \in \mathcal{I}$.

Fix any $i \in \mathcal{I}$. Since $\beta^* \in \mathcal{E}(\mathcal{R})$, it is sufficient to show that $\beta_i^*(E) \subseteq \mathcal{R}_i^{\beta^*}(\theta')$. Let $E' = \bigcup_{\theta \in E} P(\theta)$. Since $\beta_i^*(E) \subseteq \beta_i^*(E')$, the proof is complete if we show that $\beta_i^*(E') \subseteq \beta_i^*(\theta')$. Fix any $\ell \in \mathcal{I}$. Let us proceed according to whether $\ell \in \mathcal{I}^{E'}$ or not.

Suppose that $\ell \in \mathcal{I}^{E'}$. Fix any $\bar{\theta} \in E'$. Then, there exists $\theta \in E$ such that $\bar{\theta} \in P(\theta)$. By (38), $\beta(\bar{\theta}) = \beta(\theta)$, and so $\bar{\theta} \in \beta_i(\theta)$. Since $\mathcal{I}^{E'} \subseteq \mathcal{I}^E$, it follows by our initial supposition that there exists $\hat{\theta}(\ell) \in P(\theta) \subseteq E'$ such that $L_\ell(f(\theta), \hat{\theta}(\ell)) \subseteq L_\ell(f(\theta), \theta')$. Since $\hat{\theta}(\ell) \in P(\theta)$, it follows from (38) that $\beta^*(\theta) = \beta^*(\hat{\theta}(\ell))$, and so $\hat{\theta}(\ell) \in \bigcap_{j \in \mathcal{I} \setminus \{\ell\}} \beta_j^*(\theta)$. Since $\beta^*(\theta) = \beta^*(\hat{\theta}(\ell))$, IM implies that $f(\theta) = f(\hat{\theta}(\ell))$. Since the choice of $\bar{\theta} \in E'$ was arbitrary, we have that for all $\bar{\theta} \in E'$, there exist $\theta \in E'$ and $\hat{\theta}(\ell) \in E'$ such that $\bar{\theta} \in \beta_i^*(\theta), \hat{\theta}(\ell) \in \bigcap_{j \in \mathcal{I} \setminus \{\ell\}} \beta_j^*(\theta)$ and $L_\ell(f(\hat{\theta}(\ell)), \hat{\theta}(\ell)) \subseteq L_\ell(f(\theta), \theta')$.

Suppose $\ell \notin \mathcal{I}^{E'}$. Then, there exists $\tilde{\theta} \in \Theta_{\ell}^{f} \cap E'$. By definition of E', there exists $\theta \in E$ such that $\tilde{\theta} \in P(\theta)$. It follows from (38) that $\beta^{*}(\theta) = \beta^{*}(\tilde{\theta})$, and so $\theta \in \bigcap_{j \in \mathcal{I} \setminus \{\ell\}} \beta_{j}^{*}(\tilde{\theta})$. Since $\tilde{\theta} \in \Theta_{\ell}^{f}$, it follows from (10) and the fact that $\beta^{0} \subseteq \beta^{*}$ that $\beta_{\ell}^{*}(\tilde{\theta}) = \beta_{\ell}^{*}(\theta) = \Theta$, and so $E' \subseteq \beta_{\ell}^{*}(\tilde{\theta})$. Therefore, for all $\bar{\theta} \in E'$, there exist $\tilde{\theta} \in E'$ and $\theta \in E'$ such that $\bar{\theta} \in \beta_{\ell}^{*}(\tilde{\theta})$, $\theta \in \bigcap_{j \in \mathcal{I} \setminus \{\ell\}} \beta_{j}^{*}(\tilde{\theta})$ and $\tilde{\theta} \in \Theta_{\ell}^{f}$.

Since the choice of ℓ was arbitrary, it follows from (5) that $\beta_i^*(E') \subseteq \mathcal{R}_i^{\beta^*}(\theta')$. Since the choice of i was arbitrary, we conclude that $\beta_i^*(E') \subseteq \mathcal{R}_i^{\beta^*}(\theta')$ for all $i \in \mathcal{I}$. Since $\beta^* \in \mathcal{E}(\mathcal{R})$ and since $E' = \bigcup_{\theta \in E} P(\theta)$, it follows that for all $\theta \in E$, $\beta^*(\theta) \subseteq \beta^*(\theta')$. Since f is IM, Lemma 15 implies that $\beta^*(\theta) = \beta^*(\theta')$. Thus, f satisfies part (1) of SEM^{**}.

Let us show that f satisfies part (2) of SEM^{**}. Fix any $\theta, \theta', \theta'' \in \Theta$ and any $i \in \mathcal{I}$. Assume that $\mathcal{I}^{\theta} = \{i\}$ and that $P(\theta) \neq P(\theta')$. Assume, to the contrary, that $L_i(f(\theta''), \theta'') \subseteq L_i(f(\theta), \theta')$.

Let us show that f satisfies part (2) of SEM^{**}. Fix any $\theta, \theta', \theta'' \in \Theta$ and any

 $i \in \mathcal{I}$. Assume that $\mathcal{I}^{\theta} = \{i\}$ and that $P(\theta) \neq P(\theta')$. Assume, to the contrary, that $L_i(f(\theta''), \theta'') \subseteq L_i(f(\theta), \theta')$. Since $P(\theta) \neq P(\theta')$, we have from (38) that $\beta^*(\theta) \neq \beta^*(\theta')$. We show that $\beta^*(\theta) = \beta^*(\theta')$.

Since $\mathcal{I}^{\theta} = \{i\}$, we have that $\theta \in \Theta_j^f$ for all $j \in \mathcal{I} \setminus \{i\}$. It follows from the definition of β^0 in (10) and the fact that $\beta^0 \subseteq \beta^*$ that $\beta_j^*(\theta) = \Theta$ for all $j \in \mathcal{I} \setminus \{i\}$.

Let us first show that $\beta^*(\theta) \subseteq \beta^*(\theta')$. Since $\beta^* \in \mathcal{E}(\mathcal{R})$, it is sufficient to show that $\beta^*(\theta) \subseteq \mathcal{R}^{\beta^*}(\theta')$. To this end, let $E_i = \beta_i^*(\theta)$ and $E_j = \Theta$ for all $j \in \mathcal{I} \setminus \{i\}$. Fix any $k \in \mathcal{I}$. Fix any $\ell \in \mathcal{I}$ and any $\overline{\theta} \in E_{\ell}$. We proceed according to whether $\ell = i$ or not.

• Suppose that $\ell = i$. Since $\bar{\theta} \in E_{\ell}$, it follows that $\bar{\theta} \in \beta_{\ell}^{*}(\theta)$. Then, (5) applies to player ℓ because $\theta \in E_{\ell}$, $\theta'' \in \bigcap_{j \in \mathcal{I} \setminus \{\ell\}} E_{j}$, $\bar{\theta} \in \beta_{\ell}^{*}(\theta)$, $\theta'' \in \bigcap_{j \in \mathcal{I} \setminus \{\ell\}} \beta_{j}^{*}(\theta) = \Theta$ and $L_{\ell}(f(\theta''), \theta'') \subseteq L_{\ell}(f(\theta), \theta')$.

• Suppose that $\ell \neq i$. Then, (5) applies to player ℓ because $\theta \in E_{\ell}, \theta \in \bigcap_{j \in \mathcal{I} \setminus \{\ell\}} E_j$, $\bar{\theta} \in \beta_{\ell}^*(\theta), \theta \in \bigcap_{j \in \mathcal{I} \setminus \{\ell\}} \beta_j^*(\theta) \text{ and } \theta \in \Theta_{\ell}^f$.

Since this holds for any $\ell \in \mathcal{I}$ and any $\bar{\theta} \in E_{\ell}$, it follows from (5) that $\beta_k^*(\theta) \subseteq \mathcal{R}_k^{\beta^*}(\theta')$. Since the choice of $k \in \mathcal{I}$ was arbitrary, we conclude that $\beta^*(\theta) \subseteq \mathcal{R}^{\beta^*}(\theta') = \beta^*(\theta')$. Since f is IM, Lemma 15 implies that $\beta^*(\theta) = \beta^*(\theta')$, we conclude that f is SEM^{**}.

Suppose that f is SEM^{**}. Then, there exists $P \in \mathcal{P}_f$ satisfying parts (1)-(2) of the condition. We show that f satisfies IM. To this end, for all $i \in \mathcal{I}$, let $\beta_i : \Theta \mapsto 2^{\Theta} \setminus \{\emptyset\}$ be defined, for all $\theta \in \Theta$, by

$$\beta_{i}(\theta) = \begin{cases} P(\theta) & \text{if } i \in \mathcal{I}^{P(\theta)} \\ \Theta & \text{otherwise.} \end{cases}$$
(39)

Since $\theta \in \beta_i(\theta)$ for all $\theta \in \Theta$ and all $i \in \mathcal{I}$, it follows from (10) that $\beta^0 \subseteq \beta$, and so $\beta \in \mathcal{B}^t$. To complete the proof, we need only to show that $\beta \in \mathcal{E}(\mathcal{R})$. The reason is

that f is IM with respect to β because $P \in \mathcal{P}_f$. Moreover, since $\beta^* = \min \mathcal{E}(\mathcal{R})$, it follows that f is IM. Thus, let us show that $\beta \in \mathcal{E}(\mathcal{R})$. Since Lemma 1 implies that $\beta \subseteq \mathcal{R}(\beta)$, we show below that $\mathcal{R}(\beta) \subseteq \beta$.

Fix any $\theta' \in \Theta$ and any $i \in \mathcal{I}$. We show that $\mathcal{R}_i^{\beta}(\theta') \subseteq \beta_i(\theta')$. Suppose that there is a $E = \prod_{i \in \mathcal{I}} E_i$ such that E satisfies (5). We show that $E_i \subseteq \beta_i(\theta')$. Suppose that $i \notin \mathcal{I}^{P(\theta')}$. It follows from (39) that that $\beta_i(\theta') = \Theta$, and so $E_i \subseteq \beta_i(\theta')$.

Otherwise, assume that $i \in \mathcal{I}^{P(\theta')}$. Since $E_i \subseteq \mathcal{R}_i^\beta(\theta')$, it follows that there exists $E_{-i} \in (2^{\Theta} \setminus \{\emptyset\})^{I-1}$ such that (5) is satisfied. To apply SEM^{**}, we need to construct the set \overline{E} , which is defined by

$$\bar{E} = \bigcup_{i \in \mathcal{I}} \left\{ \begin{array}{c} \theta \in E_i \\ \\ \end{array} \middle| \begin{array}{c} \text{there exist } \theta' \in E_i \text{ and } \hat{\theta} \in \bigcap_{j \in \mathcal{I} \setminus \{i\}} E_j \\ \\ \text{such that } \theta' \in \beta_i \left(\theta\right) \text{ and } \hat{\theta} \in \bigcap_{j \in \mathcal{I} \setminus \{i\}} \beta_j \left(\theta\right). \end{array} \right\}$$

It can be shown that for every $i \in \mathcal{I}^{\cup_{\theta \in \bar{E}} P(\theta)}$, the set E_i satisfying (5) can be replaced with $E'_i = \bigcup_{\theta \in \bar{E}} P(\theta)$.²⁵ In what follows, for all $i \in \mathcal{I}$, let us define

$$E'_{i} = \begin{cases} \cup_{\theta \in \bar{E}} P(\theta) & \text{if } i \in \mathcal{I}^{\cup_{\theta \in \bar{E}} P(\theta)} \\ E_{i} & \text{otherwise.} \end{cases}$$

²⁵To see it, observe that for every $i \in \mathcal{I}^{\bigcup_{\theta \in \bar{E}} P(\theta)}$, it holds that $P(\theta) \cap E_i \neq \emptyset$ for all $\theta \in \bar{E}$ and that for each $\theta' \in E_i$, there exists $\theta \in \bar{E}$ such that $\theta' \in P(\theta)$. These observations follow from the fact that $\beta_i(\theta) = P(\theta)$ for all $\theta \in \bar{E}$. Next, let us show that $E'_i = \bigcup_{\theta \in E_i} P(\theta)$. Fix any $\theta' \in P(\theta)$ for some $\theta \in E_i$. Since $\theta \in E_i$, there exists $\tilde{\theta} \in \bar{E}$ such that $\theta \in P\left(\tilde{\theta}\right)$. Since $P \in P_f$ and since $\theta' \in P(\theta)$ and $\theta \in P\left(\tilde{\theta}\right)$, it can be seen that $\theta' \in E'_i$. For the converse, fix any $\theta' \in P(\theta)$ for some $\theta \in \bar{E}$. Since $\theta \in \bar{E}$, we have that there exists $\tilde{\theta} \in P(\theta) \cap E_i$. Since $P \in P_f$ and since $\theta' \in P(\theta)$ and $\tilde{\theta} \in P(\theta) \cap E_i$, it follows that $\theta' \in P\left(\tilde{\theta}\right)$ for some $\tilde{\theta} \in E_i$. Thus, $E'_i = \bigcup_{\theta \in E_i} P(\theta)$. To see that the set E_i satisfying (5) can be replaced with E'_i , fix any $\bar{\theta} \in E'_i$. Then, there exists $\theta \in E_i$ such that $\bar{\theta} \in P(\theta)$. Since $E_i \subseteq \mathbb{R}^{\beta}(\theta')$, it follows that there exist $\tilde{\theta} \in E_i$ and $\hat{\theta} \in \sum_{i=1}^{n} E_i$ such that $\bar{\theta} \in P(\theta)$. Since $E_i \subseteq \mathbb{R}^{\beta}(\theta')$, it follows that there exist $\tilde{\theta} \in E_i$ and $\theta \in \sum_{i=1}^{n} E_i$ such that $\bar{\theta} \in P(\theta)$. Since $E_i \subseteq \mathbb{R}^{\beta}(\theta')$, it follows that there exist $\tilde{\theta} \in E_i$ and $\theta \in \sum_{i=1}^{n} E_i$ such that $\bar{\theta} \in \beta_i\left(\tilde{\theta}\right)$, $\hat{\theta} \in \sum_{i=1}^{n} \beta_i\left(\hat{\theta}\right)$ and either $\tilde{\theta} \in \Theta^i_i$ or $L_i\left(f\left(\hat{\theta}\right), \hat{\theta}\right) \subseteq L_i\left(f\left(\tilde{\theta}\right), \theta'\right)$. Since the choice of $\bar{\theta} \in E'_i$ was arbitrary, we can conclude that the set E_i can be replaced with the set E_i .

Fix any $(\ell, \theta) \in \mathcal{I}^{\cup_{\theta \in \bar{E}} P(\theta)} \times \bar{E}$. Since $\ell \in \mathcal{I}^{\cup_{\theta \in \bar{E}} P(\theta)}$, it follows from (5) that for every $\theta \in E'_{\ell}$, there exist $\tilde{\theta} \in E'_{\ell}$ and $\hat{\theta} \in \bigcap_{j \in \mathcal{I} \setminus \{\ell\}} E'_{j}$ such that $\theta \in \beta_{\ell}(\tilde{\theta}), \ \hat{\theta} \in \bigcap_{j \in \mathcal{I} \setminus \{\ell\}} \beta_{j}\left(\tilde{\theta}\right)$ and

$$L_{\ell}\left(f\left(\hat{\theta}\right),\hat{\theta}\right) \subseteq L_{\ell}\left(f\left(\tilde{\theta}\right),\theta'\right) \tag{40}$$

Since, by (39), $\beta_{\ell}\left(\tilde{\theta}\right) = P(\tilde{\theta})$, we have that $\theta \in P(\tilde{\theta})$. Since $P \in \mathcal{P}_{f}$, it holds that $P(\theta) = P\left(\tilde{\theta}\right)$ and $f(\theta) = f\left(\tilde{\theta}\right)$.

We proceed according to whether $P\left(\tilde{\theta}\right) = P\left(\theta'\right)$ or not. Suppose that $P\left(\tilde{\theta}\right) = P\left(\theta'\right)$. Then, $\theta \in P\left(\theta'\right)$, and so $P\left(\theta\right) = P\left(\theta'\right)$. Since $P \in \mathcal{P}_f$, it follows that $f\left(\theta\right) = f\left(\theta'\right)$. Thus, we have that $\theta' \in P\left(\theta\right)$ such that $L_\ell\left(f\left(\theta\right), \theta'\right) \subseteq L_\ell\left(f\left(\theta\right), \theta'\right)$. Suppose that $P\left(\tilde{\theta}\right) \neq P\left(\theta'\right)$. Part (2) of SEM** implies that $\hat{\theta} \in P\left(\tilde{\theta}\right)$.²⁶ Since $P \in \mathcal{P}_f$ and since, moreover, $P\left(\theta\right) = P\left(\tilde{\theta}\right)$, it follows that $\hat{\theta} \in P\left(\theta\right)$ and $f\left(\theta\right) = f\left(\tilde{\theta}\right) = f\left(\hat{\theta}\right)$. Since (40) holds, we have there exists $\hat{\theta} \in P\left(\theta\right)$ such that $L_\ell\left(f\left(\theta\right), \hat{\theta}\right) \subseteq L_\ell\left(f\left(\theta\right), \theta'\right)$.

Since the choice of $(\ell, \theta) \in \mathcal{I}^{\cup_{\theta \in \bar{E}} P(\theta)} \times \bar{E}$ was arbitrary, we have fulfilled the premises of part (1) of SME^{**} for the pair (θ', \bar{E}) . SME^{**} implies that $P(\theta') = \bigcup_{\theta \in \bar{E}} P(\theta)$.

Since $i \in \mathcal{I}^{P(\theta')} = \mathcal{I}^{\cup_{\theta \in \bar{E}} P(\theta)}$, it follows from (39) that $\beta_i(\theta') = P(\theta')$ and that $\beta_i(\theta) = P(\theta)$ for all $\theta \in \bar{E}$. Since $P(\theta') = \bigcup_{\theta \in \bar{E}} P(\theta)$, it follows that $\bigcup_{\theta \in \bar{E}} \beta_i(\theta) = \bigcup_{\theta \in \bar{E}} P(\theta) \subseteq \beta_i(\theta')$. Since $i \in \mathcal{I}^{\cup_{\theta \in \bar{E}} P(\theta)}$, it follows that $E'_i = \bigcup_{\theta \in \bar{E}} P(\theta)$. Since $E_i \subseteq E'_i$ and $\bigcup_{\theta \in \bar{E}} P(\theta) \subseteq \beta_i(\theta')$, it follows that $E_i \subseteq \beta_i(\theta')$.

Since the choice of player $i \in \mathcal{I}$ and the choice of the set E_i were arbitrary, we have that $\mathcal{R}_i^{\beta}(\theta') \subseteq \beta_i(\theta')$ for all $i \in \mathcal{I}$. Since the choice of θ' was arbitrary, we have that $\mathcal{R}(\beta) \subseteq \beta$. Thus, $\beta \in \mathcal{E}(\mathcal{R})$.

²⁶To see it, suppose $\hat{\theta} \notin P\left(\tilde{\theta}\right)$. Since $\hat{\theta} \in \bigcap_{j \in \mathcal{I} \setminus \{\ell\}} \beta_j\left(\hat{\theta}\right)$, it follows that $\mathcal{I}^{P\left(\tilde{\theta}\right)} = \{\ell\}$. Since $\mathcal{I}^{P\left(\tilde{\theta}\right)} = \{\ell\}$ and $P\left(\tilde{\theta}\right) \neq P\left(\theta'\right)$, Part (2) of SEM^{**} implies that $L_\ell\left(f\left(\hat{\theta}\right), \hat{\theta}\right) \cap SU_\ell\left(f\left(\tilde{\theta}\right), \theta'\right) \neq \emptyset$, which contradicts (40).

D. TWO PLAYER EXAMPLE OF SECTION V

Let us reconsider the two-player example presented in Section V. For all $\bar{\theta} \in \{\theta, \theta', \theta''\}$, the set $A_i(\bar{\theta})$ is defined in (2). We see that $A_1(\theta) = A_2(\theta') = \{\theta, \theta'\}$,

 $A_1(\theta') = \{\theta'\}, A_2(\theta) = \{\theta\}, \text{ and } A_1(\theta'') = A_2(\theta'') = \{\theta''\}.$ Note that $A_1(\theta) = \{\theta, \theta'\}$ because for each lottery $z \in L_1(f(\theta), \theta) \bigcap L_2(f(\theta'), \theta')$, it holds that $u_1(z, \theta) = u_2(z, \theta') = 0.^{27}$

In what follows, we show that the SCF f presented in that example is rationalizably implementable. By Theorem 1, it suffices to show that that f is IM. We do it by setting $\beta_i^* = A_i$ for all i = 1, 2 and by showing that $\beta^* = \mathcal{R}(\beta^*)$. To this end, suppose that $\beta_i^*(\bar{\theta}) = A_i(\bar{\theta})$ for all $\bar{\theta} \in \{\theta, \theta', \theta''\}$ and all i = 1, 2. Since $\beta_i^0 \subseteq \beta_i^*$ for every player i = 1, 2, it follows that $\beta^* \in \mathcal{B}^t$. It is also clear that f satisfies IM. Thus, we are only left to show that for every player $i = 1, 2, \beta_i^*(\bar{\theta}) = \mathcal{R}_i^{\beta^*}(\bar{\theta})$ for all $\bar{\theta} \in \{\theta, \theta', \theta''\}$. Since Lemma 1 implies that $\beta^* \subseteq \mathcal{R}(\beta^*)$, we need only to show that for every player $i = 1, 2, \mathcal{R}_i^{\beta^*}(\bar{\theta}) \subseteq \beta_i^*(\bar{\theta})$ for all $\bar{\theta} \in \{\theta, \theta', \theta''\}$. To this end, fix any $\bar{\theta} \in \{\theta, \theta', \theta''\}$ and any $\hat{\theta} \in \mathcal{R}_i^{\beta^*}(\bar{\theta})$. Assume, to the contrary, that $\hat{\theta} \notin \beta_i^*(\bar{\theta})$. This implies that $\hat{\theta} \notin A_i(\bar{\theta}) = \beta_i^0(\bar{\theta})$. Since $\hat{\theta} \in \mathcal{R}_i^{\beta^*}(\bar{\theta})$, it follows that there exists $E_i \subseteq \mathcal{R}_i^{\beta^*}(\bar{\theta})$ such that $\hat{\theta} \in E_i$. (5) implies that there exists $E_{-i} \subseteq \mathcal{R}_i^{\beta^*}(\bar{\theta})$ such that, given that $\hat{\theta} \in E_i$, there exists $(\tilde{\theta}, \theta_{-i}) \in E_i \times E_{-i}$ such that $(\hat{\theta}, \theta_{-i}) \in \beta_i^*(\tilde{\theta}) \times \beta_{-i}^*(\tilde{\theta})$ and either $\tilde{\theta} \in \Theta_i^f$ or $L_i(f(\theta_{-i}), \theta_{-i}) \subseteq L_i(f(\tilde{\theta}), \bar{\theta})$. Since f satisfies NWA and $\hat{\theta} \in E_i$, $\tilde{\theta}$ and have that there exists $(\tilde{\theta}, \theta_{-i}) \in E_i \times E_{-i}$ such that $(\hat{\theta}, \theta_{-i}) \in \beta_i^*(\tilde{\theta}) \times \beta_{-i}^*(\tilde{\theta})$ and

$$L_{i}\left(f\left(\theta_{-i}\right),\theta_{-i}\right)\subseteq L_{i}\left(f\left(\tilde{\theta}\right),\bar{\theta}\right).$$
(41)

We proceed according to whether $\bar{\theta} = \theta$, $\bar{\theta} = \theta'$ or $\bar{\theta} = \theta''$. Let z be a lottery that assigns $\frac{1}{2}$ to the pure outcome a and $\frac{1}{2}$ to the pure outcome c, and let z' be a lottery that assigns $\frac{1}{2}$ to the pure outcome c and $\frac{1}{2}$ to the pure outcome d.

 $[\]overline{{}^{27}\text{To see it, take any } z \in L_1(f(\theta), \theta) \cap L_2(f(\theta'), \theta')}. \text{ Then, } u_1(z, \theta) \leq u_1(f(\theta), \theta) = 0 \text{ and } u_2(z, \theta') \leq u_2(f(\theta'), \theta') = 0. \text{ Since } u_1(\cdot, \theta) = -u_2(\cdot, \theta'), \text{ it follows that } u_1(z, \theta) + u_2(z, \theta') = 0. \text{ Since } u_1(z, \theta) \leq 0 \text{ and } u_2(z, \theta') \leq 0, \text{ we have that } u_1(z, \theta) = u_2(z, \theta') = 0.$

Case 1: $\bar{\theta} = \theta$

Suppose that i = 1. Since $\hat{\theta} \notin \beta_1^*(\theta) = A_1(\theta)$, it follows that $\hat{\theta} = \theta''$. Since $\hat{\theta} \in \beta_1^*(\tilde{\theta})$ and $\hat{\theta} = \theta''$ and since $\theta_{-1} \in \beta_{-1}^*(\tilde{\theta})$, we have that $\theta'' = \hat{\theta} = \tilde{\theta} = \theta_{-1}$. However, by construction, $e \in L_1(f(\theta''), \theta'') \cap SU_1(f(\theta''), \theta)$, which is in contradiction to (41), where $\theta'' = \tilde{\theta} = \theta_{-1}$ and $\bar{\theta} = \theta$. Thus, $\mathcal{R}_1^{\beta^*}(\theta) = \beta_1^*(\theta)$.

Suppose that i = 2. Since $\mathcal{R}_{1}^{\beta^{*}}(\theta) = \beta_{1}^{*}(\theta)$, it follows that $\theta_{-2} \in \{\theta, \theta'\}$. Moreover, since $\theta_{-2} \in \beta_{-2}^{*}(\tilde{\theta}) \cap \{\theta, \theta'\}$, it holds that $\tilde{\theta} \neq \theta''$. Since $\hat{\theta} \neq \theta$, $\hat{\theta} \in \beta_{2}^{*}(\tilde{\theta})$ and $\beta_{2}^{*}(\theta) = \{\theta\}$, we also have that $\tilde{\theta} \neq \theta$. Therefore, $\tilde{\theta} \neq \theta''$ and $\tilde{\theta} \neq \theta$, and so $\tilde{\theta} = \theta'$. Then, $\hat{\theta} \in \beta_{2}^{*}(\theta')$ and $\theta_{-2} \in \beta_{-2}^{*}(\theta')$. Since $\theta_{-2} \in \{\theta, \theta'\} \cap \beta_{-2}^{*}(\theta')$, it follows that $\theta_{-2} = \theta'$. Therefore, $\tilde{\theta} = \theta_{-2} = \theta'$. However, by construction, $z \in$ $L_{2}(f(\theta'), \theta') \cap SU_{2}(f(\theta'), \theta)$, which is in contradiction to (41), where $\tilde{\theta} = \theta_{-2} = \theta'$ and $\bar{\theta} = \theta$. Thus, $\mathcal{R}_{2}^{\beta^{*}}(\theta) = \beta_{2}^{*}(\theta)$.

Case 2: $\bar{\theta} = \theta'$

Suppose that i = 2. Since $\hat{\theta} \notin \beta_2^*(\theta') = A_2(\theta')$, it follows that $\hat{\theta} = \theta''$. Since $\hat{\theta} \in \beta_2^*(\tilde{\theta})$ and $\theta_{-2} \in \beta_{-2}^*(\tilde{\theta})$, we have that $\hat{\theta} = \tilde{\theta} = \theta_{-2} = \theta''$. By construction, $z' \in L_2(f(\theta''), \theta'') \bigcap SU_2(f(\theta''), \theta')$, which is in contradiction to (41), where $\tilde{\theta} = \theta_{-2} = \theta''$ and $\bar{\theta} = \theta'$. Thus, $\mathcal{R}_2^{\beta^*}(\theta') = \beta_2^*(\theta')$.

Suppose that i = 1. Since $\mathcal{R}_{2}^{\beta^{*}}(\theta') = \beta_{2}^{*}(\theta')$, it follows that $\theta_{-1} \in \{\theta, \theta'\}$. Moreover, since $\theta_{-1} \in \beta_{-1}^{*}(\tilde{\theta}) \cap \{\theta, \theta'\}$, it holds that $\tilde{\theta} \neq \theta''$. Since $\hat{\theta} \neq \theta'$, $\hat{\theta} \in \beta_{1}^{*}(\tilde{\theta})$ and $\beta_{1}^{*}(\theta') = \{\theta'\}$, we also have that $\tilde{\theta} \neq \theta'$. Therefore, $\tilde{\theta} \notin \{\theta', \theta''\}$, and so $\tilde{\theta} = \theta$. Then, $\hat{\theta} \in \beta_{1}^{*}(\theta)$ and $\theta_{-1} \in \beta_{-1}^{*}(\theta)$. Since $\theta_{-1} \in \{\theta, \theta'\} \cap \beta_{-1}^{*}(\theta)$, it also follows that $\theta_{-1} = \theta$. Therefore, $\tilde{\theta} = \theta_{-1} = \theta$. By construction, $z \in L_{1}(f(\theta), \theta) \cap SU_{1}(f(\theta), \theta')$, which is in contradiction to (41), where $\tilde{\theta} = \theta_{-1} = \theta$ and $\bar{\theta} = \theta'$. Thus, $\mathcal{R}_{1}^{\beta^{*}}(\theta') = \beta_{1}^{*}(\theta')$.

Case 3: $\bar{\theta} = \theta''$

To derive a contradiction, it suffices to show that $\hat{\theta} = \theta''$. Assume, to the contrary, that $\hat{\theta} \neq \theta''$. Since $\hat{\theta} \in \beta_i^*(\tilde{\theta})$, it holds that $\tilde{\theta} \neq \theta''$. Thus, $\hat{\theta}, \tilde{\theta} \in \{\theta, \theta'\}$.

Suppose that $\hat{\theta} = \theta$ and i = 1. Since $\theta \in \beta_1^*(\tilde{\theta})$, we have that $\tilde{\theta} \neq \theta'$. Since

 $\tilde{\theta} \in \{\theta, \theta'\}$, it follows that $\tilde{\theta} = \theta$. Since $\theta_{-1} \in \beta_{-1}^*(\theta)$, we have that $\hat{\theta} = \tilde{\theta} = \theta_{-1} = \theta$. However, by construction, $z \in L_1(f(\theta), \theta) \cap SU_1(f(\theta), \theta'')$, which is in contradiction to (41), where $\tilde{\theta} = \theta_{-1} = \theta$ and $\bar{\theta} = \theta''$. Thus, $\theta \notin \mathcal{R}_1^{\beta^*}(\theta'')$.

Suppose that $\hat{\theta} = \theta$ and i = 2. Then, $\theta \in \beta_2^*(\tilde{\theta})$ and $\theta_{-2} \in \beta_{-2}^*(\tilde{\theta})$. Since $\theta \notin \mathcal{R}_1^{\beta^*}(\theta'')$, it follows that $\theta_{-2} \in \mathcal{R}_{-2}^{\beta^*}(\theta'') \cap \{\theta', \theta''\}$. It cannot be that $\theta_{-2} = \theta''$ because this implies that $\tilde{\theta} = \theta''$, which is a contradiction. Therefore, it must be that $\theta_{-2} = \theta'$. Since $\theta' \in \beta_{-2}^*(\tilde{\theta})$, we have that $\tilde{\theta} = \theta'$. Thus, $\tilde{\theta} = \theta_{-2} = \theta'$. However, by construction, $z \in L_2(f(\theta'), \theta') \cap SU_2(f(\theta'), \theta'')$, which is in contradiction to (41), where $\tilde{\theta} = \theta_{-2} = \theta'$ and $\bar{\theta} = \theta''$. Thus, $\theta \notin \mathcal{R}_2^{\beta^*}(\theta'')$.

Suppose that $\hat{\theta} = \theta'$ and i = 2. Then, $\theta' \in \beta_2^*(\tilde{\theta})$ and $\theta_{-2} \in \beta_{-2}^*(\tilde{\theta})$. Since $\tilde{\theta} \in \{\theta, \theta'\}$ and $\theta' \in \beta_2^*(\tilde{\theta})$, we have that $\tilde{\theta} \neq \theta$. Thus, $\tilde{\theta} = \theta'$. Since $\theta_{-2} \in \beta_{-2}^*(\theta')$, it holds that $\theta_{-2} = \theta'$. Therefore, $\hat{\theta} = \tilde{\theta} = \theta_{-2} = \theta'$. However, by construction, $z \in L_2(f(\theta'), \theta') \cap SU_2(f(\theta'), \theta'')$, which is in contradiction to (41), where $\tilde{\theta} = \theta_{-2} = \theta'$ and $\bar{\theta} = \theta''$. Thus, $\theta' \notin \mathcal{R}_2^{\beta^*}(\theta'')$.

Suppose that $\hat{\theta} = \theta'$ and i = 1. Then, $\theta' \in \beta_1^* \left(\tilde{\theta} \right)$ and $\theta_{-1} \in \beta_{-1}^* \left(\tilde{\theta} \right)$. Since $\theta' \notin \mathcal{R}_2^{\beta^*} \left(\theta'' \right)$ and $\theta \notin \mathcal{R}_2^{\beta^*} \left(\theta'' \right)$, it follows that $\theta_{-1} = \theta''$. It follows from $\theta_{-1} \in \beta_{-1}^* \left(\tilde{\theta} \right)$ that $\tilde{\theta} = \theta''$, which is a contradiction. Thus, $\theta' \notin \mathcal{R}_1^{\beta^*} \left(\theta'' \right)$.

Thus, we conclude that $\beta^* = \mathcal{R}(\beta^*)$ for the case that $\beta_i^* = A_i$ for all i = 1, 2, and so f is rationalizably implementable.

References

- Abreu, D. and Matsushima, H. (1992). Virtual implementation in iteratively undominated strategies II: Incomplete information. Princeton University working paper.
- Bergemann, D., Morris, S., and Tercieux, O. (2011). Rationalizable implementation. Journal of Economic Theory, 146(3):1253–1274.
- Bernheim, B. D. (1984). Rationalizable strategic behavior. Econometrica: Journal of the Econometric Society, pages 1007–1028.

- Brandenburger, A. and Dekel, E. (1987). Rationalizability and correlated equilibria. Econometrica: Journal of the Econometric Society, pages 1391–1402.
- Dutta, B. and Sen, A. (1991). A necessary and sufficient condition for two-person nash implementation. *The Review of Economic Studies*, 58(1):121–128.
- Echenique, F. (2005). A short and constructive proof of tarski's fixed-point theorem. International Journal of Game Theory, 33(2):215–218.
- Jackson, M. O. (1991). Bayesian implementation. Econometrica: Journal of the Econometric Society, pages 461–477.
- Jain, R., Korpella, V., and Lombardi, M. (2021). An iterative approach to rationalizable implementation. Technical report, IEAS Working Paper No. 21-A001, Institute of Economics, Academia Sinica, Taipei, Taiwan.
- Lipman, B. L. (1994). A note on the implications of common knowledge of rationality. Games and Economic Behavior, 6(1):114–129.
- Maskin, E. (1999). Nash equilibrium and welfare optimality. The Review of Economic Studies, 66(1):23–38.
- Maskin, E. and Sjöström, T. (2002). Implementation theory. *Handbook of Social Choice and Welfare*, 1:237–288.
- Mezzetti, C. and Renou, L. (2017). Repeated nash implementation. Theoretical Economics, 12(1):249–285.
- Moore, J. and Repullo, R. (1990). Nash implementation: a full characterization. Econometrica: Journal of the Econometric Society, pages 1083–1099.
- Osborne, M. J. and Rubinstein, A. (1994). A course in game theory. MIT press.
- Oury, M. and Tercieux, O. (2012). Continuous implementation. *Econometrica*, 80(4):1605–1637.

- Pearce, D. G. (1984). Rationalizable strategic behavior and the problem of perfection. Econometrica: Journal of the Econometric Society, pages 1029–1050.
- Weinstein, J. and Yildiz, M. (2007). A structure theorem for rationalizability with application to robust predictions of refinements. *Econometrica*, 75(2):365–400.
- Xiong, S. (2022). Rationalizable implementation: Social choice function. Theoretical Economics, Forthcoming.